INTRODUCED BY COUNCIL MEMBER DANIELSON AND SECONDED FOR ADOPTION BY COUNCIL MEMBER

RESOLUTION NO. 23-08

A RESOLUTION OF THE CITY COUNCIL OF THE CITY OF MANDEVILLE
AUTHORIZING THE MAYOR OF THE CITY OF MANDEVILLE TO EXECUTE
A PROFESSIONAL SERVICES AGREEMENT BETWEEN THE CITY OF
MANDEVILLE AND WAGGONER & BALL, LLC AND PROVIDING FOR OTHER
MATTERS IN CONNECTION THEREWITH

WHEREAS, the City desires to enter into a professional services agreement with Waggoner & Ball, LLC for professional architectural services for the Old Mandeville Flood Protection Project ("Project"). The scope of the Project is for Waggoner & Ball to perform a recreational master plan for the City; and

WHEREAS, the contract is attached and made a part of this Resolution; and

NOW, THEREFORE, BE IT RESOLVED that the City Council of the City of Mandeville, hereby authorizes and empowers the Mayor of the City to execute a professional services agreement with Waggoner & Ball, LLC for architectural services on behalf of the City of Mandeville.

With the above resolution having been properly introduced and duly seconded, the vote was as follows:

AYES: NAYS: ABSTENTIONS: ABSENT:		
and the resolution was declared adopted this	day of March, 2023.	
Kristine Scherer Clerk of Council	Rick Danielson Council Chairman	

NON-CONVICTION AFFIDAVIT

STATE	OF L	OUISIANA	
PARISH	OF	XXXXX	

THE UNDERSIGNED AUTHORITY, BEFORE ME. PERSONALLY CAME AND APPEARED David W. Huchel WHO AFTER BEING BY ME DULY SWORN, DEPOSED AND SAID OP Waggonner & Ball, LLC THAT HE IS THE FULLY AUTHORISED Secretary of Sole Member (HEREIN TO AS THE REFERRED BIDDER) **PARTY** WHO **SUBMITTED** BID BID NO. Not Applicable FOR Mandeville Recreation Master Plan AND SAID AFFIANT FURTHER SAID:

He/she personally has not been convicted of, nor has he/she entered a plea of guilty or noio contendere to any of the crimes or equivalent federal crimes listed below. No individual partner, incorporator, director, manager, officer, organizer, or member, who has a minimum of a ten percent ownership in the bidding entity, has been convicted of, or has entered a plea of guilty or noio contendere to any of the crimes or equivalent federal crimes listed below.

A conviction of or plea of guilty or nolo contendere to the following state crimes or equivalent federal crimes shall permanently bar any person or the bidding entity from bidding on public projects:

- (a) Public bribery (R.S. 14:118).
- (b) Corrupt influencing (R.S. 14:120).
- (c) Extortion (R.S. 14:66).
- (d) Money laundering (R.S. 14:230).

A conviction of or plea of guilty or nolo contendere to the following state crimes or equivalent federal crimes shall bar any person or the bidding entity from bidding on public projects for a period of five years from the date of conviction or from the date of the entrance of the plea of guilty or nolo contendere:

- (a) Theft (R.S. 14:67).
- (b) Identity Theft (R.S. 14:67.16).
- (c) Theft of a business record (R.S. 14:67.20).
- (d) False accounting (R.S. 14:70).
- (e) Issuing worthless checks (R.S. 14:71).
- (f) Bank fraud (R.S. 14:71.1).
- (g) Forgery (R.S. 14:72).
- (h) Contractors; misapplication of payments (R.S. 14:202).
- (i) Malfeasance in office (R.S. 14:134).

The five-year prohibition provided for in this section shall apply only if the crime was committed during the solicitation or execution of a contract or bid awarded pursuant to these provisions. If evidence is submitted substantiating that a false attestation has been made and the project must be readvertised or the contract cancelled, the awarded entity making the false attestation shall be responsible to the public entity for the costs of rebidding, additional costs due to increased costs of bids and any and all delay costs due to the rebid or cancellation of this project.

AUTHORIZED SIGNATURE

David W. Huchel, Secretary of Sole Member

SWORN TO AND SUBSCRIBED

BEFORE ME THIS _____

DAY OF ______ 20____

See attached Certificate

NOTARY PUBLIC _____ day of _____, 20__.

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document.
State of California County of Los Angeles
Subscribed and sworn to (or affirmed) before me on this 13th
day of February , 20 23 , by David W. Huchel
proved to me on the basis of satisfactory evidence to be the person(s) who appeared before me.
KIMI SUZUKI Notary Public - California Los Angeles County Commission # 2424031 My Comm. Expires Oct 28, 1026 (Seal) Signature

Exhibit A

Mandeville Recreational Master Plan Scope of Services & Fee Proposal February 13, 2023

Prime Consultant: Waggonner & Ball (WB)
Sub Consultant: KVS Architecture (KVS)

Scope Proposal:

- 1. Discover \$30,000.00
 - a. Parks & Rec "Atlas"
 - i. Citywide maps to identify properties
 - ii. Identify data gaps
 - b. Planning Context
 - i. Summary of existing facilities & green space
 - ii. Summary of existing governance
 - iii. Summary of existing funding
 - iv. Existing planning efforts underway
 - 1. Pelican Park Master Plan
 - c. Public Engagement Summary
 - i. Online survey
 - ii. Public meetings (# TBD)
 - iii. Printed materials
 - 1. Meetings
 - 2. Mailings
 - iv. Focus groups (# TBD) for guided discussion
 - v. Strategy for engaging across language and tech barriers
 - d. Inventory & Conditions Assessment of Parks & Facilities
 - i. Park/facility type & purpose
 - ii. Location
 - iii. Area
 - iv. Inventory of facilities & amenities
 - v. General assessment of conditions & functionality
 - vi. Residence access within 10 min walk

2. Define - \$20,000.00

- a. Goals & Values Definition
- b. Needs Assessment & Analysis
 - i. Public Engagement Summary
 - ii. GIS resources & citywide maps
 - iii. Existing master plans & studies
 - iv. Inventory & Conditions Assessment
 - v. Existing demographic data
 - vi. Report Card based on NRPA park access standard

3. Design - \$15,000.00

- a. Design Workshop(s)
 - i. Identify opportunities & constraints
 - ii. Identify alignments with Resilience Plan and other efforts
- b. Design Progress Presentation
 - i. Internal to City
 - ii. Public-facing if needed
- c. NOTE: During this phase, WB will identify synergies with WB's existing work as resilience plan advisor to the City of Mandeville under a separate contract.

4. Deliver - \$15,000.00

- a. Master Plan Recommendations
 - i. Land acquisition & dispossession
 - ii. Construction/redevelopment of new facilities
 - iii. Development of additional amenities
 - iv. New programming types
- b. Project priorities to better inform future capital requests
- c. Preliminary opinions of probably cost for recommendations
 - i. High level opinions of probable cost to support future capital requests
 - ii. Preliminary maintenance budget considerations and scope

Fee Proposal:

Waggonner & Ball proposes to complete this scope for a lump sum fee of \$80,000.00, with KVS Architecture as a sub-consultant. Work to be invoiced monthly by percent complete by Phase.

Phase	Fee Per Phase
1 Discover	\$30,000.00
2 Define	\$20,000.00
3 Design	\$15,000.00
4 Deliver	\$15,000.00

TOTAL \$80,000.00

Exhibit B

WB Current Rates as of 02/13/22

Rate Schedule

Principal	\$341	/hr
Senior Architect	\$226	/hr
Senior Project Designer	\$184	/hr
Architect	\$173	/hr
Senior Designer	\$142	/hr
Designer	\$116	/hr

Charges for outside services, equipment, and facilities not furnished directly by Waggonner & Ball will be billed at cost plus 15%. Such charges may include, but shall not be limited to, printing reproduction services; shipping, delivery, and courier charges; consumable materials; subconsultant fees and expenses; special fees, permits and insurance; transportation and accommodations. Mileage will be charged at the prevailing IRS rate per mile.



Standard Abbreviated Form of Agreement Between Owner and Architect

AGREEMENT made as of the day of in the year 2023 (In words, indicate day, month and year.)

BETWEEN the Architect's client identified as the Owner: (Name, legal status, address and other information)

City of Mandeville 3101 E Causeway Approach Mandeville, LA 70448

and the Architect:

(Name, legal status, address and other information)

Waggonner & Ball, LLC 2200 Prytania Street New Orleans, LA 70130

for the following Project: (Name, location and detailed description)

Mandeville Recreational Master Plan Mandeville, Louisiana

The Owner and Architect agree as follows.

ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

User Notes:

(1147368257)

TABLE OF ARTICLES

- 1 INITIAL INFORMATION
- 2 ARCHITECT'S RESPONSIBILITIES
- 3 SCOPE OF ARCHITECT'S BASIC SERVICES
- 4 SUPPLEMENTAL AND ADDITIONAL SERVICES
- 5 OWNER'S RESPONSIBILITIES
- 6 COST OF THE WORK
- 7 COPYRIGHTS AND LICENSES
- 8 CLAIMS AND DISPUTES
- 9 TERMINATION OR SUSPENSION
- 10 MISCELLANEOUS PROVISIONS
- 11 COMPENSATION
- 12 SCOPE OF THE AGREEMENT

ARTICLE 1 INITIAL INFORMATION

§ 1.1 This Agreement is based on the Initial Information set forth below:

(State below details of the Project's site and program, Owner's contractors and consultants, Architect's consultants, Owner's budget for the Cost of the Work, and other information relevant to the Project.)

See Exhibit A – Scope of Services and Fee Proposal for project scope and fee.

- § 1.2 The Owner and Architect may rely on the Initial Information. Both parties, however, recognize that such information may materially change and, in that event, the Owner and the Architect shall appropriately adjust the schedule, the Architect's services and the Architect's compensation. The Owner shall adjust the Owner's budget for the Cost of the Work and the Owner's anticipated design and construction milestones, as necessary, to accommodate material changes in the Initial Information.
- § 1.3 The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form

(Paragraph deleted)

Init.

ARTICLE 2 ARCHITECT'S RESPONSIBILITIES

- § 2.1 The Architect shall provide the professional services set forth in this Agreement consistent with the professional skill and care ordinarily provided by architects practicing in the same or similar locality under the same or similar circumstances. The Architect shall perform its services as expeditiously as is consistent with such professional skill and care and the orderly progress of the Project.
- § 2.2 The Architect shall maintain the following insurance until termination of this Agreement. If any of the requirements set forth below are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect as set forth in Section 11.5:

(Identify types and limits of insurance coverage, and other insurance requirements applicable to the Agreement, if any.)

.1 General Liability

\$1,000,000 per occurrence/\$2,000,000 policy aggregate

.2 Automobile Liability

\$1,000,000 limit

.3 Workers' Compensation

\$1,000,000

.4 Professional Liability

\$2,000,000 per claim \$4,000,000 policy aggregate

- § 2.2.5 The Architect shall ensure that the Owner is listed as an additional insured under each policy of insurance described in this Section, excluding Worker's Compensation and Professional Liability Insurance. Furthermore, there shall be a waiver of subrogation in favor of Owner.
- § 2.2.6 All certificates of insurance shall be furnished to the Owner, and said insurance shall not be canceled without thirty (30) days prior notice of cancellation given to the Owner, in writing. Architect shall provide proof of liability and workers' compensation insurance to Owner (Said policies and Declaration sheets to be delivered to Owner before commencement of work performed under this Agreement.)
- § 2.3 To the fullest extent permitted by law, Architect will indemnify and hold Owner, its officials, employees, and agents harmless from and against: (1) any and all losses, damages or liabilities, suits, judgments of sums of money to any party accruing against Owner for loss of life or injury or damage to persons or property but only to the extent growing out of, resulting from, or by reason of any negligent act or omission or the operation of the Architect or any of its subcontractors, or any of its or their agents, servants, employees, while engaged in or about or in connection with the discharge or performance of the services to be done or performed by the Architect under this Agreement; and (2) any and all claims and/or liens for labor, services, or materials furnished to the Architect in connection with the performance of its obligation under this Agreement.

ARTICLE 3 SCOPE OF ARCHITECT'S BASIC SERVICES

- § 3.1 The Architect's Basic Services consist of those described in this Article 3 and include usual and customary structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.
- § 3.1.1 The Architect shall coordinate its services with those services provided by the Owner and the Owner's consultants. The Architect shall be entitled to rely on (1) the accuracy and completeness of the services and information furnished by the Owner and (2) the Owner's approvals. The Architect shall provide prompt written notice to the Owner if the Architect becomes aware of any error, omission, or inconsistency in such services or information.
- § 3.1.2 As soon as practicable after the date of this Agreement, the Architect shall submit for the Owner's approval a schedule for the performance of the Architect's services. Once approved by the Owner, time limits established by the schedule shall not, except for reasonable cause, be exceeded by the Architect or Owner. With the Owner's approval, the Architect shall adjust the schedule, if necessary, as the Project proceeds until the commencement of construction.
- § 3.1.3 The Architect shall assist the Owner in connection with the Owner's responsibility for filing documents required for the approval of governmental authorities having jurisdiction over the Project.

§ 3.2 Design Phase Services

§ 3.2.1 The Architect shall review the program and other information furnished by the Owner, and shall review laws, codes, and regulations applicable to the Architect's services.

- § 3.2.2 The Architect shall discuss with the Owner the Owner's program, schedule, budget for the Cost of the Work, Project site, and alternative approaches to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the Project requirements.
- § 3.2.3 The Architect shall consider the relative value of alternative materials, building systems and equipment, together with other considerations based on program, aesthetics, and any sustainable objectives, in developing a design for the Project that is consistent with the Owner's schedule and budget for the Cost of the Work.
- § 3.2.4 Based on the Project requirements, the Architect shall prepare Design Documents for the Owner's approval consisting of drawings and other documents appropriate for the Project and the Architect shall prepare and submit to the Owner an estimate of the Cost of the Work prepared in accordance with Section 6.3.
- § 3.2.5 The Architect shall submit the Design Documents to the Owner, and request the Owner's approval.

§ 3.3 Construction Documents Phase Services

- § 3.3.1 Based on the Owner's approval of the Design Documents, the Architect shall prepare for the Owner's approval Construction Documents consisting of Drawings and Specifications setting forth in detail the requirements for the construction of the Work. The Owner and Architect acknowledge that in order to construct the Work the Contractor will provide additional information, including Shop Drawings, Product Data, Samples and other similar submittals, which the Architect shall review in accordance with Section 3.4.4.
- § 3.3.2 The Architect shall incorporate the design requirements of governmental authorities having jurisdiction over the Project into the Construction Documents.
- § 3.3.3 The Architect shall submit the Construction Documents to the Owner, update the estimate for the Cost of the Work and advise the Owner of any adjustments to the estimate of the Cost of the Work, take any action required under Section 6.5, and request the Owner's approval.
- § 3.3.4 The Architect, following the Owner's approval of the Construction Documents and of the latest estimate of the Cost of the Work, shall assist the Owner in obtaining bids or proposals and awarding and preparing contracts for construction.

§ 3.4 Construction Phase Services

§ 3.4.1 General

- § 3.4.1.1 The Architect shall provide administration of the Contract between the Owner and the Contractor as set forth below and in AIA Document A104TM—2017, Standard Abbreviated Form of Agreement Between Owner and Contractor. If the Owner and Contractor modify AIA Document A104—2017, those modifications shall not affect the Architect's services under this Agreement unless the Owner and the Architect amend this Agreement.
- § 3.4.1.2 The Architect shall advise and consult with the Owner during the Construction Phase Services. The Architect shall have authority to act on behalf of the Owner only to the extent provided in this Agreement. The Architect shall not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, nor shall the Architect be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect shall be responsible for the Architect's negligent acts or omissions, but shall not have control over or charge of and shall not be responsible for, acts or omissions of the Contractor or of any other persons or entities performing portions of the Work.
- § 3.4.1.3 Subject to Section 4.1, the Architect's responsibility to provide Construction Phase Services commences with the award of the Contract for Construction and terminates on the date the Architect issues the final Certificate for Payment.

§ 3.4.2 Evaluations of the Work

Init.

§ 3.4.2.1 The Architect shall visit the site at intervals appropriate to the stage of construction, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the

Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.

- § 3.4.2.2 The Architect has the authority to reject Work that does not conform to the Contract Documents and has the authority to require inspection or testing of the Work.
- § 3.4.2.3 The Architect shall interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests shall be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- § 3.4.2.4 When making such interpretations and decisions, the Architect shall endeavor to secure faithful performance by both Owner and Contractor, shall not show partiality to either, and shall not be liable for results of interpretations or decisions rendered in good faith.
- § 3.4.2.5 The Architect shall render initial decisions on Claims between the Owner and Contractor as provided in the Contract Documents.

§ 3.4.3 Certificates for Payment to Contractor

- § 3.4.3.1 The Architect shall review and certify the amounts due the Contractor and shall issue certificates in such amounts. The Architect's certification for payment shall constitute a representation to the Owner, based on the Architect's evaluation of the Work as provided in Section 3.4.2 and on the data comprising the Contractor's Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified.
- § 3.4.3.2 The issuance of a Certificate for Payment shall not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) ascertained how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

§ 3.4.4 Submittals

- § 3.4.4.1 The Architect shall review and approve, or take other appropriate action, upon the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or any construction means, methods, techniques, sequences or procedures.
- § 3.4.4.2 If the Contract Documents specifically require the Contractor to provide professional design services or certifications by a design professional related to systems, materials or equipment, the Architect shall specify the appropriate performance and design criteria that such services must satisfy. The Architect shall review and take appropriate action on Shop Drawings and other submittals related to the Work designed or certified by the Contractor's design professional, provided the submittals bear such professional's seal and signature when submitted to the Architect. The review shall be for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect shall be entitled to rely upon, and shall not be responsible for, the adequacy and accuracy of the services, certifications, and approvals performed or provided by such design professionals.

§ 3.4.4.3 The Architect shall review and respond to written requests for information about the Contract Documents. The Architect's response to such requests shall be made in writing within any time limits agreed upon, or otherwise with reasonable promptness.

§ 3.4.5 Changes in the Work

The Architect may order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. Subject to Section 4.1.2, the Architect shall prepare Change Orders and Construction Change Directives for the Owner's approval and execution in accordance with the Contract Documents.

§ 3.4.6 Project Completion

The Architect shall conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion; forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract Documents and received from the Contractor; and issue a final Certificate for Payment based upon a final inspection indicating that, to the best of the Architect's knowledge, information, and belief, the Work complies with the requirements of the Contract Documents.

ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES § 4.1

(Paragraphs deleted)

The Architect may provide Additional Services after execution of this Agreement without invalidating the Agreement. Upon recognizing the need to perform Additional Services, the Architect shall notify the Owner. The Architect shall not provide the Additional Services until the Architect receives the Owner's written authorization. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 4.1 shall entitle the Architect to compensation pursuant to Section 11.3.

(Paragraphs deleted)

- § 4.1.1 The Architect shall provide services necessitated by a change in the Initial Information, changes in previous instructions or approvals given by the Owner, or a material change in the Project including size; quality; complexity; the Owner's schedule or budget for Cost of the Work; or procurement or delivery method as an Additional Service.
- § 4.1.2 The Architect shall, as an Additional Service, provide services made necessary by a Contractor's proposed change in the Work. The Architect shall prepare revisions to the Architect's Instruments of Service necessitated by Change Orders and Construction Change Directives as an Additional Service.

ARTICLE 5 OWNER'S RESPONSIBILITIES

- § 5.1 Unless otherwise provided for under this Agreement, the Owner shall provide information in a timely manner regarding requirements for and limitations on the Project, including a written program which shall set forth the Owner's objectives, schedule, constraints and criteria, including space requirements and relationships, flexibility, expandability, special equipment, systems and site requirements.
- § 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.
- § 5.3 The Owner shall furnish surveys to describe physical characteristics, legal limitations and utility locations for the site of the Project; a written legal description of the site; and services of geotechnical engineers or other consultants, when the Architect requests such services and demonstrates that they are reasonably required by the scope of the Project.
- § 5.4 The Owner shall coordinate the services of its own consultants with those services provided by the Architect. Upon the Architect's request, the Owner shall furnish copies of the scope of services in the contracts between the

User Notes:

(1147368257)

Owner and the Owner's consultants. The Owner shall require that its consultants and contractors maintain insurance, including professional liability insurance, as appropriate to the services or work provided.

- § 5.5 The Owner shall furnish tests, inspections and reports required by law or the Contract Documents, such as structural, mechanical, and chemical tests; tests for air and water pollution; and tests for hazardous materials.
- § 5.6 The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.
- § 5.7 The Owner shall provide prompt written notice to the Architect if the Owner becomes aware of any fault or defect in the Project, including errors, omissions or inconsistencies in the Architect's Instruments of Service.
- § 5.8 The Owner shall endeavor to communicate with the Contractor through the Architect about matters arising out of or relating to the Contract Documents.
- § 5.9 The Owner shall provide the Architect access to the Project site prior to commencement of the Work and shall obligate the Contractor to provide the Architect access to the Work wherever it is in preparation or progress.
- § 5.10 Within 15 days after receipt of a written request from the Architect, the Owner shall furnish the requested information as necessary and relevant for the Architect to evaluate, give notice of, or enforce lien rights.

ARTICLE 6 COST OF THE WORK

- § 6.1 For purposes of this Agreement, the Cost of the Work shall be the total cost to the Owner to construct all elements of the Project designed or specified by the Architect and shall include contractors' general conditions costs, overhead and profit. The Cost of the Work also includes the reasonable value of labor, materials, and equipment, donated to, or otherwise furnished by, the Owner. The Cost of the Work does not include the compensation of the Architect; the costs of the land, rights-of-way, financing, or contingencies for changes in the Work; or other costs that are the responsibility of the Owner.
- § 6.2 The Owner's budget for the Cost of the Work is provided in Initial Information, and shall be adjusted throughout the Project as required under Sections 5.2, 6.4 and 6.5. Evaluations of the Owner's budget for the Cost of the Work, and the preliminary estimate of the Cost of the Work and updated estimates of the Cost of the Work prepared by the Architect, represent the Architect's judgment as a design professional. It is recognized, however, that neither the Architect nor the Owner has control over the cost of labor, materials or equipment; the Contractor's methods of determining bid prices; or competitive bidding, market or negotiating conditions. Accordingly, the Architect cannot and does not warrant or represent that bids or negotiated prices will not vary from the Owner's budget for the Cost of the Work, or from any estimate of the Cost of the Work, or evaluation, prepared or agreed to by the Architect.
- § 6.3 In preparing estimates of the Cost of Work, the Architect shall be permitted to include contingencies for design, bidding and price escalation; to determine what materials, equipment, component systems and types of construction are to be included in the Contract Documents; to recommend reasonable adjustments in the program and scope of the Project; and to include design alternates as may be necessary to adjust the estimated Cost of the Work to meet the Owner's budget. The Architect's estimate of the Cost of the Work shall be based on current area, volume or similar conceptual estimating techniques. If the Owner requires a detailed estimate of the Cost of the Work, the Architect shall provide such an estimate, if identified as the Architect's responsibility in Section 4.1, as a Supplemental Service.
- § 6.4 If, through no fault of the Architect, construction procurement activities have not commenced within 90 days after the Architect submits the Construction Documents to the Owner the Owner's budget for the Cost of the Work shall be adjusted to reflect changes in the general level of prices in the applicable construction market.
- § 6.5 If at any time the Architect's estimate of the Cost of the Work exceeds the Owner's budget for the Cost of the Work, the Architect shall make appropriate recommendations to the Owner to adjust the Project's size, quality or budget for the Cost of the Work, and the Owner shall cooperate with the Architect in making such adjustments.
- § 6.6 If the Owner's current budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services is exceeded by the lowest bona fide bid or negotiated proposal, the Owner shall

- .1 give written approval of an increase in the budget for the C of the Work;
- .2 authorize rebidding or renegotiating of the Project within a reasonable time;
- .3 terminate in accordance with Section 9.5;
- .4 in consultation with the Architect, revise the Project program, scope, or quality as required to reduce the Cost of the Work; or
- .5 implement any other mutually acceptable alternative.

§ 6.7 If the Owner chooses to proceed under Section 6.6.4, the Architect shall modify the Construction Documents as necessary to comply with the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services, or the budget as adjusted under Section 6.6.1. If the Owner requires the Architect to modify the Construction Documents because the lowest bona fide bid or negotiated proposal exceeds the Owner's budget for the Cost of the Work due to market conditions the Architect could not reasonably anticipate, the Owner shall compensate the Architect for the modifications as an Additional Service pursuant to Section 11.3; otherwise the Architect's services shall be without additional compensation. In any event, the Architect's modification of the Construction Documents shall be the limit of the Architect's responsibility under this Article 6.B

ARTICLE 7 COPYRIGHTS AND LICENSES

- § 7.1 The Architect and the Owner warrant that in transmitting Instruments of Service, or any other information, the transmitting party is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project.
- § 7.2 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements or for similar purposes in connection with the Project is not to be construed as publication in derogation of the reserved rights of the Architect and the Architect's consultants.
- § 7.3 The Architect grants to the Owner a nonexclusive license to use the Architect's Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations under this Agreement, including prompt payment of all sums when due pursuant to Article 9 and Article 11. The Architect shall obtain similar nonexclusive licenses from the Architect's consultants consistent with this Agreement. The license granted under this section permits the Owner to authorize the Contractor, Subcontractors, Sub-subcontractors, and suppliers, as well as the Owner's consultants and separate contractors, to reproduce applicable portions of the Instruments of Service, subject to any protocols established pursuant to Section 1.3, solely and exclusively for use in performing services or construction for the Project. If the Architect rightfully terminates this Agreement for cause as provided in Section 9.4, the license granted in this Section 7.3 shall terminate.
- § 7.3.1 In the event the Owner uses the Instruments of Service without retaining the authors of the Instruments of Service, the Owner releases the Architect and Architect's consultant(s) from all claims and causes of action arising from such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the Architect and its consultants from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's use of the Instruments of Service under this Section 7.3.1. The terms of this Section 7.3.1 shall not apply if the Owner rightfully terminates this Agreement for cause under Section 9.4.
- § 7.4 Except for the licenses granted in this Article 7, no other license or right shall be deemed granted or implied under this Agreement. The Owner shall not assign, delegate, sublicense, pledge or otherwise transfer any license granted herein to another party without the prior written agreement of the Architect. Any unauthorized use of the Instruments of Service shall be at the Owner's sole risk and without liability to the Architect and the Architect's consultants.
- § 7.5 Except as otherwise stated in Section 7.3, the provisions of this Article 7 shall survive the termination of this Agreement.

١

ARTICLE 8 CLAIMS AND DISPUTES

§ 8.1 General

- § 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1.
- § 8.1.2 To the extent damages are covered by property insurance, the Owner and Architect waive all rights against each other and against the contractors, consultants, agents, and employees of the other, for damages, except such rights as they may have to the proceeds of such insurance as set forth in AIA Document A104–2017, Standard Abbreviated Form of Agreement Between Owner and Contractor. The Owner or the Architect, as appropriate, shall require of the contractors, consultants, agents, and employees of any of them, similar waivers in favor of the other parties enumerated herein.
- § 8.1.3 The Architect and Owner waive consequential damages for claims, disputes or other matters in question, arising out of or relating to this Agreement. This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination of this Agreement, except as specifically provided in Section 9.6.

§ 8.2 Mediation

- § 8.2.1 Any claim, dispute or other matter in question arising out of or related to this Agreement shall be subject to mediation as a condition precedent to binding dispute resolution. If such matter relates to or is the subject of a lien arising out of the Architect's services, the Architect may proceed in accordance with applicable law to comply with the lien notice or filing deadlines prior to resolution of the matter by mediation or by binding dispute resolution.
- § 8.2.2 Mediation, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of this Agreement. The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.
- § 8.2.3 If the parties do not resolve a dispute through mediation pursuant to this Section 8.2, the method of binding dispute resolution shall be the following: (Check the appropriate box.)

-]	Arbitration pursuant to Section 8.3 of this Agreement
X]	Litigation in a court of competent jurisdiction
1	Other: (Specify)

If the Owner and Architect do not select a method of binding dispute resolution, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, the dispute will be resolved in a court of competent jurisdiction.

(Paragraphs deleted)

§ 8.2.4 The provisions of this Article 8 shall survive the termination of this Agreement.

ARTICLE 9 TERMINATION OR SUSPENSION

§ 9.1 If the Owner fails to make payments to the Architect in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Architect's option, cause for suspension of performance of services under this Agreement. If the Architect elects to suspend services, the Architect shall give seven days' written notice to the Owner before suspending services. In the event of a suspension of services, the Architect shall have no liability to the Owner for delay or damage caused the Owner because of such suspension of

services. Before resuming services, the Owner shall pay the Architect all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

- § 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.3 If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Architect, the Architect may terminate this Agreement by giving not less than seven days' written notice.
- § 9.4 Either party may terminate this Agreement upon not less than seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.
- § 9.5 The Owner may terminate this Agreement upon not less than seven days' written notice to the Architect for the Owner's convenience and without cause.
- § 9.6 In the event of termination not the fault of the Architect, the Architect shall be compensated for services performed prior to termination, Reimbursable Expenses incurred, and all costs attributable to termination, including the costs attributable to the Architect's termination of consultant agreements.

§ 9.7

(Paragraphs deleted)

Except as otherwise expressly provided herein, this Agreement shall terminate one year from the date of Substantial Completion.

(Paragraph deleted)

ARTICLE 10 MISCELLANEOUS PROVISIONS

- § 10.1 This Agreement shall be governed by the law of the place where the Project is located excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 8.3.
- § 10.2 Terms in this Agreement shall have the same meaning as those in AIA Document A104–2017, Standard Abbreviated Form of Agreement Between Owner and Contractor.
- § 10.3 The Owner and Architect, respectively, bind themselves, their agents, successors, assigns and legal representatives to this Agreement. Neither the Owner nor the Architect shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement, including any payments due to the Architect by the Owner prior to the assignment.
- § 10.4 If the Owner requests the Architect to execute certificates or consents, the proposed language of such certificates or consents shall be submitted to the Architect for review at least 14 days prior to the requested dates of execution. The Architect shall not be required to execute certificates or consents that would require knowledge, services or responsibilities beyond the scope of this Agreement.
- § 10.5 Nothing contained in this Agreement shall create a contractual relationship with, or a cause of action in favor of, a third party against either the Owner or Architect.
- § 10.6 The Architect shall have no responsibility for the discovery, presence, handling, removal or disposal of, or exposure of persons to, hazardous materials or toxic substances in any form at the Project site.
- § 10.7 The Architect shall have the right to include photographic or artistic representations of the design of the Project among the Architect's promotional and professional materials. However, the Architect's materials shall not include

information the Owner has identified in writing as confidential or proprietary. The Owner shall provide professional credit for the Architect in the Owner's promotional materials for the Project. This Section 10.7 shall survive the termination of this Agreement unless the Owner terminates this Agreement for cause pursuant to Section 9.4.

- § 10.8 The invalidity of any provision of the Agreement shall not invalidate the Agreement or its remaining provisions. If it is determined that any provision of the Agreement violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Agreement shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Agreement.
- § 10.9 Neither Architect nor anyone employed by Architect shall be considered an employee of the Owner for the purpose of unemployment compensation coverage, the same being hereby expressly waived by the parties hereto.
- § 10.10 The continuation of this Agreement is contingent upon the appropriation of funds to fulfill the requirements of the Agreement by the Owner. If the Owner fails to appropriate sufficient monies to provide for the continuation of this Agreement, or if such appropriation is reduced by any means provided in the appropriations act, or for any other lawful purpose, and the effect of such reduction is to provide insufficient monies for the continuation of the Agreement, this Agreement shall terminate on the date of the beginning of the first fiscal year for which funds are not appropriated.

ARTICLE 11 COMPENSATION

- § 11.1 For the Architect's Basic Services described under Article 3, the Owner shall compensate the Architect as follows:
 - .1 Stipulated Sum (Insert amount)

\$80,000.00 Lump Sum by Phase, See Exhibit A

(Paragraphs deleted)

- § 11.2 For Supplemental Services identified in Section 4.1, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation. If necessary, list specific services to which particular methods of compensation apply.)
- § 11.3 For Additional Services that may arise during the course of the Project, including those under Section 4.1, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation.)

Compensation will be determined prior to the commencement of work.

§ 11.4 The hourly billing rates for services of the Architect and the Architect's consultants, if any, are set forth below. The rates shall be adjusted in accordance with the Architect's and Architect's consultants' normal review practices. (If applicable, attach an exhibit of hourly billing rates or insert them below.)

See Exhibit B

§ 11.5

User Notes:

(Paragraphs deleted)

Compensation for Reimbursable Expenses

§ 11.5.1 Reimbursable Expenses are in addition to compensation for Basic, Supplemental, and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:

Init.

- .1 Transportation and authorized out-of-town travel and subsistence;
- .2 Long distance services, dedicated data and communication services, teleconferences, Project web sites, and extranets;
- .3 Permitting and other fees required by authorities having jurisdiction over the Project;
- .4 Printing, reproductions, plots, and standard form documents:
- .5 Postage, handling, and delivery:
- .6 Expense of overtime work requiring higher than regular rates if authorized in advance by the Owner;
- .7 Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner or required for the Project;
- .8 Expense of professional liability insurance dedicated exclusively to this Project or the expense of additional insurance coverage or limits requested by the Owner in excess of that normally maintained by the Architect and the Architect's consultants;
- .9 All taxes levied on professional services and on reimbursable expenses;
- .10 Site office expenses; and
- .11 Other similar Project-related expenditures.
- § 11.5.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus fifteen percent (15 %) of the expenses incurred.

§ 11.6 Payments to the Architect

§ 11.6.1 Initial Payment

An initial payment of zero (\$ 0) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.6.2 Progress Payments

§ 11.6.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid sixty (60) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect.

(Insert rate of monthly or annual interest agreed upon.)

6 % Per Annum

- § 11.6.2.2 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.
- § 11.6.2.3 Records of Reimbursable Expenses, expenses pertaining to Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

ARTICLE 12 SCOPE OF THE AGREEMENT

- § 12.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both the Owner and Architect.
- § 12.2 This Agreement is comprised of the following documents identified below:
 - .1 AIA Document B104™—2017, Standard Abbreviated Form of Agreement Between Owner and Architect
 - .3 Exhibits:

(Clearly identify any other exhibits incorporated into this Agreement, including any exhibits identified in Section 4.1.)

Exhibit A - Scope of Services & Fee Proposal Exhibit B - WB Current Rates as of 02/13/22

This Agreement entered into as of the day a	nd year first written above.
	gelf Stant
OWNER (Signature)	ARCHITECT (Signature)
Clay Madden Mayor	Andrew J. Sternad, Business Unit Leader
(Printed name and title)	(Printed name title and license number if required)

Additions and Deletions Report for

AIA® Document B104® - 2017

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 20:31:46 ET on 02/13/2023.

PAGE 1

AGREEMENT made as of the day of in the year 2023

City of Mandeville 3101 E Causeway Approach Mandeville, LA 70448

Waggonner & Ball, LLC 2200 Prytania Street New Orleans, LA 70130

Mandeville Recreational Master Plan Mandeville, Louisiana PAGE 2

12 SPECIAL TERMS AND CONDITIONS

-SCOPE OF THE AGREEMENT

User Notes:

<u>See Exhibit A – Scope of Services and Fee Proposal for project scope and fee.</u>

§ 1.3 The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. The parties will use AIA Document E203™ 2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.form

§ 1.3.1 Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in AIA Document E203™ 2013, Building Information Modeling and Digital Data Exhibit, and the requisite AIA Document G202™ 2013, Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

Additions and Deletions Report for AIA Document B104 - 2017. Copyright © 1974, 1978, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects, "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks of The American Institute of Architects. This document was produced at 20:31:46 ET on 02/13/2023 under Order No.2114305826 which expires on 04/01/2023, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

(1147368257)

§ 2.2 The Architect shall maintain the following insurance until termination of this Agreement. If any of the requirements set forth below are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect as set forth in Section 11.8:11.5:

PAGE 3

\$1,000,000 per occurrence/\$2,000,000 policy aggregate

\$1,000,000 limit

\$1,000,000

- .4 Professional Liability
- \$2,000,000 per claim \$4,000,000 policy aggregate
- § 2.2.5 The Architect shall ensure that the Owner is listed as an additional insured under each policy of insurance described in this Section, excluding Worker's Compensation and Professional Liability Insurance. Furthermore, there shall be a waiver of subrogation in favor of Owner.
- § 2.2.6 All certificates of insurance shall be furnished to the Owner, and said insurance shall not be canceled without thirty (30) days prior notice of cancellation given to the Owner, in writing. Architect shall provide proof of liability and workers' compensation insurance to Owner (Said policies and Declaration sheets to be delivered to Owner before commencement of work performed under this Agreement.)
- § 2.3 To the fullest extent permitted by law, Architect will indemnify and hold Owner, its officials, employees, and agents harmless from and against: (1) any and all losses, damages or liabilities, suits, judgments of sums of money to any party accruing against Owner for loss of life or injury or damage to persons or property but only to the extent growing out of, resulting from, or by reason of any negligent act or omission or the operation of the Architect or any of its subcontractors, or any of its or their agents, servants, employees, while engaged in or about or in connection with the discharge or performance of the services to be done or performed by the Architect under this Agreement; and (2) any and all claims and/or liens for labor, services, or materials furnished to the Architect in connection with the performance of its obligation under this Agreement.

 PAGE 4
- § 3.4.1.3 Subject to Section 4.2, 4.1, the Architect's responsibility to provide Construction Phase Services commences with the award of the Contract for Construction and terminates on the date the Architect issues the final Certificate for Payment.
- § 3.4.2.1 The Architect shall visit the site at intervals appropriate to the stage of construction, or as otherwise required in Section 4.2.2, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.

PAGE 6

The Architect may order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. Subject to Section 4.2.3, 4.1.2, the Architect shall prepare Change Orders and Construction Change Directives for the Owner's approval and execution in accordance with the Contract Documents.

...

§ 4.1 Supplemental Services are not included in Basic Services but may be required for the Project. The Architect shall provide the Supplemental Services indicated below, and the Owner shall compensate the Architect as provided in Section 11.2. Supplemental Services may include programming, site evaluation and planning, environmental studies, civil engineering, landscape design, telecommunications/data, security, measured drawings of existing conditions, coordination of separate contractors or independent consultants, detailed cost estimates, on site project representation beyond requirements of Section 4.2.2, value analysis, interior architectural design, tenant related services, preparation of record drawings, commissioning, sustainable project services, and any other services not otherwise included in this Agreement. (Identify below the Supplemental Services that the Architect is required to provide and insert a description of each Supplemental Service, if not further described in an exhibit attached to this document.)

The Architect may provide Additional Services after execution of this Agreement without invalidating the Agreement. Upon recognizing the need to perform Additional Services, the Architect shall notify the Owner. The Architect shall not provide the Additional Services until the Architect receives the Owner's written authorization. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 4.1 shall entitle the Architect to compensation pursuant to Section 11.3.

- § 4.2 The Architect may provide Additional Services after execution of this Agreement without invalidating the Agreement. Upon recognizing the need to perform Additional Services, the Architect shall notify the Owner. The Architect shall not provide the Additional Services until the Architect receives the Owner's written authorization. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 1.2 shall entitle the Architect to compensation pursuant to Section 1.1.3.
- § 4.2.1 The Architect shall provide services necessitated by a change in the Initial Information, changes in previous instructions or approvals given by the Owner, or a material change in the Project including size; quality; complexity; the Owner's schedule or budget for Cost of the Work; or procurement or delivery method as an Additional Service.
- § 4.2.2 The Architect has included in Basic Services () visits to the site by the Architect during construction. The Architect shall conduct site visits in excess of that amount as an Additional Service.
- § 4.2.3 The Architect shall, as an Additional Service, provide services made necessary by a Contractor's proposed change in the Work. The Architect shall prepare revisions to the Architect's Instruments of Service necessitated by Change Orders and Construction Change Directives as an Additional Service.
- § 4.2.4 If the services covered by this Agreement have not been completed within—() months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.
- § 4.1.1 The Architect shall provide services necessitated by a change in the Initial Information, changes in previous instructions or approvals given by the Owner, or a material change in the Project including size; quality; complexity; the Owner's schedule or budget for Cost of the Work; or procurement or delivery method as an Additional Service.
- § 4.1.2 The Architect shall, as an Additional Service, provide services made necessary by a Contractor's proposed change in the Work. The Architect shall prepare revisions to the Architect's Instruments of Service necessitated by Change Orders and Construction Change Directives as an Additional Service.

PAGE 8

.1 give written approval of an increase in the budget for the Cost C of the Work;

•••

§ 6.7 If the Owner chooses to proceed under Section 6.6.4, the Architect shall modify the Construction Documents as necessary to comply with the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services, or the budget as adjusted under Section 6.6.1. If the Owner requires the Architect to modify the Construction Documents because the lowest bona fide bid or negotiated proposal exceeds the Owner's budget for the Cost of the Work due to market conditions the Architect could not reasonably anticipate, the Owner shall compensate the Architect for the modifications as an Additional Service pursuant to Section 11.3; otherwise the Architect's services shall be without additional compensation. In any event, the Architect's modification of the Construction Documents shall be the limit of the Architect's responsibility under this Article 6.6.B

PAGE 9

[X] Litigation in a court of competent jurisdiction

•••

§ 8.3 Arbitration

- § 8.3.1 If the parties have selected arbitration as the method for binding dispute resolution in this Agreement, any claim, dispute or other matter in question arising out of or related to this Agreement subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of the Agreement.
- § 8.3.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the claim, dispute or other matter in question would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the claim, dispute or other matter in question.
- § 8.3.2 The foregoing agreement to arbitrate, and other agreements to arbitrate with an additional person or entity duly consented to by parties to this Agreement, shall be specifically enforceable in accordance with applicable law in any court having jurisdiction thereof.
- § 8.3.3 The award rendered by the arbitrator(s) shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

§ 8.3.4 Consolidation or Joinder

- § 8.3.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation; (2) the arbitrations to be consolidated substantially involve common questions of law or fact; and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).
- § 8.3.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.
- § 8.3.4.3 The Owner and Architect grant to any person or entity made a party to an arbitration conducted under this Section 8.3, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Architect under this Agreement.
- § 8.4 The provisions of this Article 8 shall survive the termination of this Agreement.
- § 8.2.4 The provisions of this Article 8 shall survive the termination of this Agreement.

PAGE 10

§ 9.7 In addition to any amounts paid under Section 9.6, if the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall pay to the Architect the following fees:

(Set forth below the amount of any termination or licensing fee, or the method for determining any termination or licensing fee.)

- 1 Termination Fee:
- .2 Licensing Fee if the Owner intends to continue using the Architect's Instruments of Service:

Except as otherwise expressly provided herein, this Agreement shall terminate one year from the date of Substantial Completion.

§ 9.8 Except as otherwise expressly provided herein, this Agreement shall terminate one year from the date of Substantial Completion.

PAGE 11

:

- § 10.9 Neither Architect nor anyone employed by Architect shall be considered an employee of the Owner for the purpose of unemployment compensation coverage, the same being hereby expressly waived by the parties hereto.
- § 10.10 The continuation of this Agreement is contingent upon the appropriation of funds to fulfill the requirements of the Agreement by the Owner. If the Owner fails to appropriate sufficient monies to provide for the continuation of this Agreement, or if such appropriation is reduced by any means provided in the appropriations act, or for any other lawful purpose, and the effect of such reduction is to provide insufficient monies for the continuation of the Agreement, this Agreement shall terminate on the date of the beginning of the first fiscal year for which funds are not appropriated.

\$80,000.00 Lump Sum by Phase, See Exhibit A

.2	- Percentage Basis (Insert percentage value)
	() % of the Owner's budget for the Cost of the Work, as calculated in accordance with Section 11.6
.3	Other — (Describe the method of compensation)

§ 11.3 For Additional Services that may arise during the course of the Project, including those under Section 4.2, 4.1, the Owner shall compensate the Architect as follows:

Compensation will be determined prior to the commencement of work.

§ 11.4 Compensation for Supplemental and Additional Services of the Architect's consultants when not included in Section 11.2 or 11.3, shall be the amount invoiced to the Architect plus percent (%), or as follows: The hourly billing rates for services of the Architect and the Architect's consultants, if any, are set forth below. The rates shall be adjusted in accordance with the Architect's and Architect's consultants' normal review practices. (If applicable, attach an exhibit of hourly billing rates or insert them below.)

See Exhibit B

§ 11.5 Where compensation for Basic Services is based on a stipulated sum or percentage of the Cost of the Work, the compensation for each phase of services shall be as follows:

Design Phase		percent (%)
Construction Documents		percent (%)
Phase				•
Construction Phase		percent (%)
				•
Total Basic Compensation	one-hundred	percent (100	%)

Compensation for Reimbursable Expenses

§ 11.5.1 Reimbursable Expenses are in addition to compensation for Basic, Supplemental, and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:

- Transportation and authorized out-of-town travel and subsistence;
- Long distance services, dedicated data and communication services, teleconferences, Project web sites, and extranets;
- Permitting and other fees required by authorities having jurisdiction over the Project;
- Printing, reproductions, plots, and standard form documents;
- Postage, handling, and delivery;
- Expense of overtime work requiring higher than regular rates if authorized in advance by the Owner;
- Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner or required for the Project;
- Expense of professional liability insurance dedicated exclusively to this Project or the expense of additional insurance coverage or limits requested by the Owner in excess of that normally maintained by the Architect and the Architect's consultants;
- All taxes levied on professional services and on reimbursable expenses;
- .10 Site office expenses; and
- .11 Other similar Project-related expenditures.
- § 11.5.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus fifteen percent (15 %) of the expenses incurred.
- § 11.6 When compensation identified in Section 11.1 is on a percentage basis, progress payments for each phase of Basic Services shall be calculated by multiplying the percentages identified in this Article by the Owner's most recent budget for the Cost of the Work. Compensation paid in previous progress payments shall not be adjusted based on subsequent updates to the Owner's budget for the Cost of the Work.

Payments to the Architect

§ 11.6.1 When compensation is on a percentage basis and any portions of the Project are deleted or otherwise not constructed, compensation for those portions of the Project shall be payable to the extent services are performed on those portions. The Architect shall be entitled to compensation in accordance with this Agreement for all services performed whether or not the Construction Phase is commenced Initial Payment

An initial payment of zero (\$ 0) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.6.2 Progress Payments

§ 11.6.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid sixty (60) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect.

(Insert rate of monthly or annual interest agreed upon.)

6 % Per Annum

- § 11.6.2.2 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.
- § 11.6.2.3 Records of Reimbursable Expenses, expenses pertaining to Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

ARTICLE 12 SCOPE OF THE AGREEMENT

§ 12.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both the Owner and Architect.

§ 12.2 This Agreement is comprised of the following documents identified below:

.1 AIA Document B104TM—2017, Standard Abbreviated Form of Agreement Between Owner and Architect

.3 Exhibits:

(Clearly identify any other exhibits incorporated into this Agreement, including any exhibits identified in Section 4.1.)

Exhibit A - Scope of Services & Fee Proposal
Exhibit B - WB Current Rates as of 02/13/22

(If applicable, attach an exhibit of hourly billing rates or insert them below.)

This Agreement entered into as of the day and year first written above.

OWNER (Signature)	ARCHITECT (Signature)
Clay Madden Mayor	Andrew J. Sternad, Business Unit Leader
(Printed name and title)	(Printed name, title, and license number, if required)
OWNER (Signature)	ARCHITECT (Signature)
(Printed name and title)	(Printed name, title, and license number, if required)
§ 11.7 The hourly billing rates for services of the Archite	et and the Architect's consultants, if any, are set forth below.
	ect's and Architect's consultants' normal ravious practices

Employee or Category

Rate

§ 11.8 Compensation for Reimbursable Expenses

- § 11.8.1 Reimbursable Expenses are in addition to compensation for Basic, Supplemental, and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:
 - .1 Transportation and authorized out-of-town-travel and subsistence;
 - Long distance services, dedicated data and communication services, teleconferences, Project web sites, and extranets;
 - 3 Permitting and other fees required by authorities having jurisdiction over the Project;

- .4 Printing, reproductions, plots, and standard form documents;
- .5 Postage, handling, and delivery;
- 6 Expense of overtime work requiring higher than regular rates if authorized in advance by the Owner;
- .7 Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner or required for the Project;
- 8 Expense of professional liability insurance dedicated exclusively to this Project or the expense of additional insurance coverage or limits requested by the Owner in excess of that normally maintained by the Architect and the Architect's consultants;
- All taxes levied on professional services and on reimbursable expenses;
- .10 Site office expenses; and
- .11 Other similar Project-related expenditures.

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus—percent (—%) of the expenses incurred.

§ 11.9 Payments to the Architect

§ 11.9.1 Initial Payment

An initial payment of (\$--) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.9.2 Progress Payments

§ 11.9.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid () days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect. (Insert rate of monthly or annual interest agreed upon.)

%

§ 11.9.2.2 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.

§ 11.9.2.3 Records of Reimbursable Expenses, expenses pertaining to Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

ARTICLE 12 SPECIAL TERMS AND CONDITIONS

Special terms and conditions that modify this Agreement are as follows: (Include other terms and conditions applicable to this Agreement.)

ARTICLE 13 SCOPE OF THE AGREEMENT

§ 13.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both the Owner and Architect.

§ 13.2 This Agreement is comprised of the following documents identified below:

- 1—AIA Document B104TM 2017, Standard Abbreviated Form of Agreement Between Owner and Architect
- AIA Document E203TM 2013, Building Information Modeling and Digital Data Exhibit, dated as indicated below:
- (Insert the date of the E203-2013 incorporated into this agreement.)

.3—	- Exhibits:
	-(Clearly identify any other exhibits incorporated into this Agreement, including any exhibits identified
	in Section 4.1.)
.4—	- Other-documents:
	-(List other documents, if any, including additional scopes of service forming part of the Agreement.)

This Agreement entered into as of the day and year first written above.

Certification of Document's Authenticity

AIA® Document D401 ™ - 2003

I, Andrew Sternad, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 20:31:46 ET on 02/13/2023 under Order No. 2114305826 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA® Document B104TM – 2017, Standard Abbreviated Form of Agreement Between Owner and Architect, other than those additions and deletions shown in the associated Additions and Deletions Report.

	all Stant
(Signea	
	Business Unit Leader
(Title)	
	February 13, 2023
(Dated)	

ATTACHMENT "A"

CODE OF CONDUCT AFFIDAVIT

contained in Louisiana Revised Statutes at Title 42, Chapter 15 and is familiar with and has a copy of the Code of Conduct for the City of Mandeville; and That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental	STATE OF LOUISIANA
BEFORE ME, the undersigned Notary Public personally came and appeared:	PARISH OF_XXXXX
David W. Huchel, Secretary of Sole Member of Waggenner & Ball, LLC. ; if a corporation, LLC, or partnership, then the affidavit may be signed by corporate officer, member or partner stating that the subject organization is aware of the Code of Governmental Ethics. Who, after first being duly sworn did depose and say he/she is the president, corporate officer, or designee of Waggonner & Ball, LLC and that he/she is familiar with the Code of Governmental Ethics contained in Louisiana Revised Statutes at Title 42, Chapter 15 and is familiar with and has a copy of the Code of Conduct for the City of Mandeville; and That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental	AFFIDAVIT
Sole Member of Waggenner & Ball, LLC; if a corporation, LLC, or partnership, then the amidavit may be signed by corporate officer, member or partner stating that the subject organization is aware of the Code of Governmental Ethics. Who, after first being duly sworn did depose and say he/she is the president, corporate officer, or designee of Waggonner & Ball, LLC	BEFORE ME, the undersigned Notary Public personally came and appeared:
Who, after first being duly swom did depose and say he/she is the president, corporate officer, or designee of Waggonner & Ball, LLC and that he/she is familiar with the Code of Governmental Bthics contained in Louisiana Revised Statutes at Title 42, Chapter 15 and is familiar with and has a copy of the Code of Conduct for the City of Mandeville; and That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental	David W. Huchel, Secretary of; if a corporation, LLC, or partnership, then the affidavit may be signed by
Who, after first being duly sworn did depose and say he/she is the president, corporate officer, or designee of Waggonner & Ball, LLC and that he/she is familiar with the Code of Governmental Bthics contained in Louisiana Revised Statutes at Title 42, Chapter 15 and is familiar with and has a copy of the Code of Conduct for the City of Mandeville; and That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental	corporate officer, member or partner stating that the subject organization is aware of the Code of
Waggonner & Ball, LLC and that he/she is familiar with the Code of Governmental Ethics contained in Louisiana Revised Statutes at Title 42, Chapter 15 and is familiar with and has a copy of the Code of Conduct for the City of Mandeville; and That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental	Governmental Ethics.
contained in Louisiana Revised Statutes at Title 42, Chapter 15 and is familiar with and has a copy of the Code of Conduct for the City of Mandeville; and That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental	Who, after first being duly sworn did depose and say he/she is the president, corporate officer, or designee of
Code of Conduct for the City of Mandeville; and That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental	Waggonner & Ball, LLC and that he/she is familiar with the Code of Governmental Ethics
That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental	contained in Louisiana Revised Statutes at Title 42, Chapter 15 and is familiar with and has a copy of the
•	Code of Conduct for the City of Mandeville; and
Taking and the Code of Conduct on a condition of any contract between the commony and the City of Mandaville	That, Waggonner & Ball, LLC will conform to the provisions in the Code of Governmental
Editics and the Code of Colleget as a condition of any conflact between the company and the City of managemen,	Ethics and the Code of Conduct as a condition of any contract between the company and the City of Mandeville,
Louisiana; and	Louisiana; and
That, the provisions of the Code of Governmental Ethics and the Code of Conduct shall apply to any	That, the provisions of the Code of Governmental Ethics and the Code of Conduct shall apply to any
subcontract that the contractor has with any other entity in the performance of its contract with the City of	subcontract that the contractor has with any other entity in the performance of its contract with the City of
Mandeville, Louisiana and there shall be a statement in every subcontract to that effect.	Mandeville, Louisiana and there shall be a statement in every subcontract to that effect.
WITNESSES: CONTRACTOR:	WITNESSES: CONTRACTOR:
By: Pauren	, ,
David W. Huchel, Secretary of Sole Member	David W. Huchel, Secretary of Sole Membe
See attached Certificate	See attached Certificate
Sworn to and subscribed before me thisday of, 2010.	Sworn to and subscribed before me thisday of, 2010.

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document.								
State of California County of Los Angeles								
Subscribed and sworn to (or affirmed) before me on this 13th day of February , 20 23 , by David W. Huchel								
proved to me on the basis of satisfactory evidence to be the person(s) who appeared before me.								
KIMI SUZUKI Notary Public - California Los Angeles County Commission # 2424031 My Comm. Expires Oct 28, 2026 Signature Kumulumulumulumulumulumulumulumulumulumu								



CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/DD/YYYY) 2/13/2023

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must have ADDITIONAL INSURED provisions or be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s).

	this certificate does not confer rights to the certificate holder in lieu of such endorsement(s).												
PRODUCER							CONTACT NAME: Karin Thorp						
AssuredPartners Design Professionals Insurance Services, LLC						PHONE							
3697 Mt. Diablo Blvd Suite 230 Lafayette CA 94549							ADDRESS: DesignProCerts@assuredpartners.com						
							INSURER(S) AFFORDING COVERAGE NAIC						
License#; 6003745											20443		
	SURED				MOFF&NI-01	INSURER B: Transportation Insurance Company					20494		
		onner & Ball, LLC						ntal Insurance			35289		
2200 Prytania St. New Orleans LA 70130									surance Co PA		19445		
The state of the s							INSURER E : XL Specialty Insurance Company						
							INSURER F:				37885		
6	COVERAGES CERTIFICATE NUMBER: 2083035819 REVISION NUMBER:												
Ť	THIS IS TO CERTIFY THAT THE POLICIES OF INSURANCE LISTED BELOW HAVE BEEN ISSUED TO THE INSURED NAMED ABOVE FOR THE POLICY PERIOD												
	INDICATED. NOTWITHSTANDING ANY REQUIREMENT, TERM OR CONDITION OF ANY CONTRACT OR OTHER DOCUMENT WITH RESPECT TO WHICH THIS												
l	CERTIFICATE MAY BE ISSUED OR MAY PERTAIN, THE INSURANCE AFFORDED BY THE POLICIES DESCRIBED HEREIN IS SUBJECT TO ALL THE TERMS, EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN REDUCED BY PAID CLAIMS.												
IN	SR R	TYPE OF INSURANCE	ADDL	SUBR			POLICY EFF (MM/DD/YYYY)	POLICY EXP (MM/DD/YYYY)	LIMIT				
	A X	COMMERCIAL GENERAL LIABILITY	INSD	WVD	6056712992		11/22/2022	11/22/2023	EACH OCCURRENCE	\$ 1,000	000		
Ι΄	` ~				00007 12002			111212020	DAMAGE TO RENTED PREMISES (Ea occurrence)				
	x									\$1,000			
	Î	BFPD XCU	1						MED EXP (Any one person)	\$ 15,00	-		
l		Ded/SIR \$0					1		PERSONAL & ADV INJURY	\$ 1,000			
l	GEN	N'L AGGREGATE LIMIT APPLIES PER:							GENERAL AGGREGATE	\$ 2,000			
	\vdash	POLICY X PRO-							PRODUCTS - COMP/OP AGG	\$ 2,000 \$ 15.00			
┝	B AUT	OTHER:	-	-	6056742000		11/22/2022	11/22/2023	Project/Location Agg COMBINED SINGLE LIMIT		<u> </u>		
l	X	ANY AUTO			6056712989		11/22/2022	11/22/2023	(Ea accident) BODILY INJURY (Per person)	\$ 1,000,000			
ŀ	^								BODILY INJURY (Per accident)	s			
l		OWNED SCHEDULED AUTOS HIRED NON-OWNED							PROPERTY DAMAGE	s			
		AUTOS ONLY AUTOS ONLY							(Per accident)	-			
H	- V		ļ		0070007000		44/00/0000	44/00/0000	Comp/Collision	\$1,000			
Ι'	×			6079627982		11/22/2022	11/22/2023	EACH OCCURRENCE \$10,00					
CENTROWADE			l						AGGREGATE	\$ 10,00	0,000		
H	D WORKERS COMPENSATION				14/04007000		44/00/0000	44/00/0000	X PER OTH-	\$			
AND EMPLOYERS' LIABILITY ANYPROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED?					WC19397906	11/22/2022	11/22/2023						
			N/A						E.L. EACH ACCIDENT \$1,000				
(Mandatory in NH)									E.L. DISEASE - EA EMPLOYEE				
If yes, describe under DESCRIPTION OF OPERATIONS below E Professional Liability			L	<u> </u>			4410010000	4.4100.40000	E.L. DISEASE - POLICY LIMIT	\$ 1,000			
	inclu	essional Liability ides Contractors Pollution	N		DPR5003484		11/22/2022	11/22/2023	Per Claim Aggregate	2,000			
l	Clair	ms-Made		İ									
L													
10	ESCRIPT	non of operations / Locations / VEHIC a policy is a follow-form to underlying	LES (#	ACORD	ı 101, Additional Remarks Schedul I Liahility/Auto Liahility/Emr	le, may b Novers	e attached if mor Liability	e space is requir	od)				
١١	Vorker'	s Compensation extends to include	Jone	s Act	and USL&H.	,,o,o,o	Liability						
[Re: WB	2217 Mandeville Recreational Mast	er Pl	an.	ne and Voluntoom are nam	nd ac a	n additional i	neurod se roe	nacte conoral and auto lis	hilih, a	e required per		
I١	The City of Mandeville, its Officers, Agents, Employees and Volunteers are named as an additional insured as respects general and auto liability as required per written contract or agreement. General and Auto Liability are Primary/Non-Contributory per policy form wording. Insurance coverage includes waiver of												
1	subrogation per the attached endorsement(s).												
ı													
CERTIFICATE HOLDER							CANCELLATION 30 Days and 10 for Non-Payment						
							SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN						
		City of Mandeville, Its Office	ers	Age	nts. Employees and				Y PROVISIONS.				
١		Volunteers			, <u></u>								
3101 E. Causeway Approach Mandeville LA 70448						AUTHORIZED REPRESENTATIVE							
IVIANUEVIIIE LA 70440							Karin Thosp						
1		•				ا السك	I word it is by						



NOTICE OF CANCELLATION TO CERTIFICATEHOLDERS

It is understood and agreed that:

If you have agreed under written contract to provide notice of cancellation to a party to whom the Agent of Record has issued a Certificate of Insurance, and if we cancel a policy term described on that Certificate of Insurance for any reason other than nonpayment of premium, then notice of cancellation will be provided to such Certificateholders at least 30 days in advance of the date cancellation is effective.

If notice is mailed, then proof of mailing to the last known mailing address of the Certificateholder on file with the Agent of Record will be sufficient to prove notice.

Any failure by us to notify such persons or organizations will not extend or invalidate such cancellation, or impose any liability or obligation upon us or the Agent of Record.



CNA PARAMOUNT

Architects, Engineers and Surveyors General Liability Extension Endorsement

by the indemnitee at the Insurer's request will be paid as **defense costs**. Such payments will not be deemed to be **damages** for **personal and advertising injury** and will not reduce the limits of insurance.

C. This PERSONAL AND ADVERTISING INJURY - LIMITED CONTRACTUAL LIABILITY Provision does not apply if Coverage B –Personal and Advertising Injury Liability is excluded by another endorsement attached to this Coverage Part.

This PERSONAL AND ADVERTISING INJURY - CONTRACTUAL LIABILITY Provision does not apply to any person or organization who otherwise qualifies as an additional insured on this Coverage Part.

22. PROPERTY DAMAGE - ELEVATORS

- A. Under COVERAGES, Coverage A Bodily Injury and Property Damage Liability, the paragraph entitled Exclusions is amended such that the Damage to Your Product Exclusion and subparagraphs (3), (4) and (6) of the Damage to Property Exclusion do not apply to property damage that results from the use of elevators.
- B. Solely for the purpose of the coverage provided by this PROPERTY DAMAGE ELEVATORS Provision, the Other Insurance conditions is amended to add the following paragraph:

This insurance is excess over any of the other insurance, whether primary, excess, contingent or on any other basis that is Property insurance covering property of others damaged from the use of elevators.

23. RETIRED PARTNERS, MEMBERS, DIRECTORS AND EMPLOYEES

WHO IS INSURED is amended to include as Insureds natural persons who are retired partners, members, directors or employees, but only for bodily injury, property damage or personal and advertising injury that results from services performed for the Named Insured under the Named Insured's direct supervision. All limitations that apply to employees and volunteer workers also apply to anyone qualifying as an Insured under this Provision.

24. SUPPLEMENTARY PAYMENTS

The section entitled SUPPLEMENTARY PAYMENTS - COVERAGES A AND B is amended as follows:

- A. Paragraph 1.b. is amended to delete the \$250 limit shown for the cost of bail bonds and replace it with a \$5,000. limit; and
- **B.** Paragraph **1.d.** is amended to delete the limit of \$250 shown for daily loss of earnings and replace it with a \$1,000. limit.

25. UNINTENTIONAL FAILURE TO DISCLOSE HAZARDS

If the Named Insured unintentionally fails to disclose all existing hazards at the inception date of the Named Insured's Coverage Part, the Insurer will not deny coverage under this Coverage Part because of such failure.

26. WAIVER OF SUBROGATION - BLANKET

Under CONDITIONS, the condition entitled Transfer Of Rights Of Recovery Against Others To Us is amended to add the following:

The Insurer waives any right of recovery the Insurer may have against any person or organization because of payments the Insurer makes for injury or damage arising out of:

- 1. the Named Insured's ongoing operations; or
- 2. your work included in the products-completed operations hazard.

However, this waiver applies only when the Named Insured has agreed in writing to waive such rights of recovery in a written contract or written agreement, and only if such contract or agreement:

CNA74858XX (1-15)
Page 16 of 18
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: 8 Effective Date: 11/22/2022





Architects, Engineers and Surveyors General Liability Extension Endorsement

- 1. is in effect or becomes effective during the term of this Coverage Part; and
- was executed prior to the bodily injury, property damage or personal and advertising injury giving rise to the claim.

27. WRAP-UP EXTENSION: OCIP, CCIP, OR CONSOLIDATED (WRAP-UP) INSURANCE PROGRAMS

Note: The following provision does not apply to any public construction project in the state of Oklahoma, nor to any construction project in the state of Alaska, that is not permitted to be insured under a **consolidated (wrap-up)** insurance program by applicable state statute or regulation.

If the endorsement **EXCLUSION – CONSTRUCTION WRAP-UP** is attached to this policy, or another exclusionary endorsement pertaining to Owner Controlled Insurance Programs (O.C.I.P.) or Contractor Controlled Insurance Programs (C.C.I.P.) is attached, then the following changes apply:

A. The following wording is added to the above-referenced endorsement:

With respect to a consolidated (wrap-up) insurance program project in which the Named Insured is or was involved, this exclusion does not apply to those sums the Named Insured become legally obligated to pay as damages because of:

- Bodily injury, property damage, or personal or advertising injury that occurs during the Named Insured's
 ongoing operations at the project, or during such operations of anyone acting on the Named Insured's
 behalf; nor
- 2. Bodily injury or property damage included within the products-completed operations hazard that arises out of those portions of the project that are not residential structures.
- B. Condition 4. Other Insurance is amended to add the following subparagraph 4.b.(1)(c):

This insurance is excess over:

- (c) Any of the other insurance whether primary, excess, contingent or any other basis that is insurance available to the Named Insured as a result of the Named Insured being a participant in a consolidated (wrap-up) insurance program, but only as respects the Named Insured's involvement in that consolidated (wrap-up) insurance program.
- C. DEFINITIONS is amended to add the following definitions:

Consolidated (wrap-up) insurance program means a construction, erection or demolition project for which the prime contractor/project manager or owner of the construction project has secured general liability insurance covering some or all of the contractors or subcontractors involved in the project, such as an Owner Controlled Insurance Program (O.C.I.P.) or Contractor Controlled Insurance Program (C.C.I.P.).

Residential structure means any structure where 30% or more of the square foot area is used or is intended to be used for human residency, including but not limited to:

- 1. single or multifamily housing, apartments, condominiums, townhouses, co-operatives or planned unit developments; and
- 2. the common areas and structures appurtenant to the structures in paragraph 1. (including pools, hot tubs, detached garages, guest houses or any similar structures).

However, when there is no individual ownership of units, **residential structure** does not include military housing, college/university housing or dormitories, long term care facilities, hotels or motels. **Residential structure** also does not include hospitals or prisons.





Blanket Additional Insured - Owners, Lessees or Contractors - with Products-Completed Operations Coverage Endorsement

This endorsement modifies insurance provided under the following:

COMMERCIAL GENERAL LIABILITY COVERAGE PART

It is understood and agreed as follows:

- I. WHO IS AN INSURED is amended to include as an Insured any person or organization whom you are required by written contract to add as an additional insured on this coverage part, but only with respect to liability for bodily injury, property damage or personal and advertising injury caused in whole or in part by your acts or omissions, or the acts or omissions of those acting on your behalf:
 - A. in the performance of your ongoing operations subject to such written contract; or
 - B. in the performance of your work subject to such written contract, but only with respect to bodily injury or property damage included in the products-completed operations hazard, and only if:
 - 1. the written contract requires you to provide the additional insured such coverage; and
 - 2. this coverage part provides such coverage.
- II. But if the written contract requires:
 - A. additional insured coverage under the 11-85 edition, 10-93 edition, or 10-01 edition of CG2010, or under the 10-01 edition of CG2037; or
 - B. additional insured coverage with "arising out of" language; or
 - C. additional insured coverage to the greatest extent permissible by law;

then paragraph I. above is deleted in its entirety and replaced by the following:

WHO IS AN INSURED is amended to include as an Insured any person or organization whom you are required by written contract to add as an additional insured on this coverage part, but only with respect to liability for bodily injury, property damage or personal and advertising injury arising out of your work that is subject to such written contract.

- III. Subject always to the terms and conditions of this policy, including the limits of insurance, the Insurer will not provide such additional insured with:
 - A. coverage broader than required by the written contract; or
 - B. a higher limit of insurance than required by the written contract.
- IV. The insurance granted by this endorsement to the additional insured does not apply to **bodily injury**, **property damage**, or **personal and advertising injury** arising out of:
 - A. the rendering of, or the failure to render, any professional architectural, engineering, or surveying services, including:
 - 1. the preparing, approving, or failing to prepare or approve maps, shop drawings, opinions, reports, surveys, field orders, change orders or drawings and specifications; and
 - 2. supervisory, inspection, architectural or engineering activities; or
 - B. any premises or work for which the additional insured is specifically listed as an additional insured on another endorsement attached to this coverage part.
- V. Under COMMERCIAL GENERAL LIABILITY CONDITIONS, the Condition entitled Other Insurance is amended to add the following, which supersedes any provision to the contrary in this Condition or elsewhere in this coverage part:

CNA75079XX (10-16) Page 1 of 2

CONTINENTAL CASUALTY COMPANY Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: Effective Date: 11/22/2022

20020008260567129825382





Blanket Additional Insured - Owners, Lessees or Contractors - with Products-Completed Operations Coverage Endorsement

Primary and Noncontributory Insurance

With respect to other insurance available to the additional insured under which the additional insured is a named insured, this insurance is primary to and will not seek contribution from such other insurance, provided that a written contract requires the insurance provided by this policy to be:

- 1. primary and non-contributing with other insurance available to the additional insured; or
- primary and to not seek contribution from any other insurance available to the additional insured.

But except as specified above, this insurance will be excess of all other insurance available to the additional insured.

VI. Solely with respect to the insurance granted by this endorsement, the section entitled **COMMERCIAL GENERAL LIABILITY CONDITIONS** is amended as follows:

The Condition entitled **Duties In The Event of Occurrence, Offense, Claim or Suit** is amended with the addition of the following:

Any additional insured pursuant to this endorsement will as soon as practicable:

- 1. give the Insurer written notice of any claim, or any occurrence or offense which may result in a claim;
- 2. send the Insurer copies of all legal papers received, and otherwise cooperate with the Insurer in the investigation, defense, or settlement of the claim; and
- 3. make available any other insurance, and tender the defense and indemnity of any **claim** to any other insurer or self-insurer, whose policy or program applies to a loss that the Insurer covers under this **coverage part**. However, if the **written contract** requires this insurance to be primary and non-contributory, this paragraph 3. does not apply to insurance on which the additional insured is a named insured.

The Insurer has no duty to defend or indemnify an additional insured under this endorsement until the Insurer receives written notice of a claim from the additional insured.

VII. Solely with respect to the insurance granted by this endorsement, the section entitled **DEFINITIONS** is amended to add the following definition:

Written contract means a written contract or written agreement that requires you to make a person or organization an additional insured on this coverage part, provided the contract or agreement:

- A. is currently in effect or becomes effective during the term of this policy; and
- B. was executed prior to:
 - 1. the bodily injury or property damage; or
 - 2. the offense that caused the personal and advertising injury;

for which the additional insured seeks coverage.

Any coverage granted by this endorsement shall apply solely to the extent permissible by law.

All other terms and conditions of the Policy remain unchanged.

This endorsement, which forms a part of and is for attachment to the Policy issued by the designated Insurers, takes effect on the effective date of said Policy at the hour stated in said Policy, unless another effective date is shown below, and expires concurrently with said Policy.

CNA75079XX (10-16)
Page 2 of 2
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: Effective Date: 11/22/2022

Copyright CNA All Rights Reserved. Includes copyrighted material of Insurance Services Office, Inc., with its permission.





It is understood and agreed that this endorsement amends the COMMERCIAL GENERAL LIABILITY COVERAGE PART as follows. If any other endorsement attached to this policy amends any provision also amended by this endorsement, then that other endorsement controls with respect to such provision, and the changes made by this endorsement with respect to such provision do not apply.

TABLE OF CONTENTS		
1.	Additional Insureds	
2.	Additional Insured - Primary And Non-Contributory To Additional Insured's Insurance	
3.	Additional Insured – Extended Coverage	
4.	Boats	
5.	Bodily Injury – Expanded Definition	
6.	Broad Knowledge of Occurrence/ Notice of Occurrence	
7.	Broad Named Insured	
8.	Contractual Liability – Railroads	
9.	Estates, Legal Representatives and Spouses	
10.	Expected Or Intended Injury – Exception for Reasonable Force	
11.	General Aggregate Limits of Insurance – Per Location	
12.	In Rem Actions	
13.	Incidental Health Care Malpractice Coverage	
14.	Joint Ventures/Partnership/Limited Liability Companies	
15.	Legal Liability – Damage To Premises	
16.	Liquor Liability	
17.	Medical Payments	
18.	Non-owned Aircraft Coverage	
19.	Non-owned Watercraft	
20.	Personal And Advertising Injury – Discrimination or Humiliation	
21.	Personal And Advertising Injury - Contractual Liability	
22.	Property Damage – Elevators	
23.	Retired Partners, Members, Directors And Employees	
24.	Supplementary Payments	
25.	Unintentional Failure To Disclose Hazards	
26.	Waiver of Subrogation – Blanket	
27.	Wrap-Up Extension: OCIP, CCIP or Consolidated (Wrap-Up) Insurance Programs	



CNA74858XX (1-15) Page 1 of 18

CONTINENTAL CASUALTY COMPANY Insured Name: MOFFATT & NICHOL

Endorsement No:

Policy No: 6056712992



Architects, Engineers and Surveyors General Liability Extension Endorsement

1. ADDITIONAL INSUREDS

- a. WHO IS AN INSURED is amended to include as an Insured any person or organization described in paragraphs A. through I. below whom a Named Insured is required to add as an additional insured on this Coverage Part under a written contract or written agreement, provided such contract or agreement:
 - (1) is currently in effect or becomes effective during the term of this Coverage Part; and
 - (2) was executed prior to:
 - (a) the bodily injury or property damage; or
 - (b) the offense that caused the personal and advertising injury,

for which such additional insured seeks coverage.

- **b.** However, subject always to the terms and conditions of this policy, including the limits of insurance, the Insurer will not provide such additional insured with:
 - (1) a higher limit of insurance than required by such contract or agreement; or
 - (2) coverage broader than required by such contract or agreement, and in no event broader than that described by the applicable paragraph A. through I. below.

Any coverage granted by this endorsement shall apply only to the extent permissible by law.

A. Controlling Interest

Any person or organization with a controlling interest in a **Named Insured**, but only with respect to such person or organization's liability for **bodily injury**, **property damage** or **personal and advertising injury** arising out of:

- 1. such person or organization's financial control of a Named Insured; or
- 2. premises such person or organization owns, maintains or controls while a **Named Insured** leases or occupies such premises;

provided that the coverage granted by this paragraph does not apply to structural alterations, new construction or demolition operations performed by, on behalf of, or for such additional insured.

B. Co-owner of Insured Premises

A co-owner of a premises co-owned by a **Named Insured** and covered under this insurance but only with respect to such co-owner's liability for **bodily injury**, **property damage** or **personal and advertising injury** as co-owner of such premises.

C. Engineers, Architects or Surveyors Engaged By You

An architect, engineer or surveyor engaged by the **Named Insured**, but only with respect to liability for **bodily injury**, **property damage** or **personal and advertising injury** caused in whole or in part by the **Named Insured's** acts or omissions, or the acts or omissions of those acting on the **Named Insured's** behalf:

- a. in connection with the Named Insured's premises; or
- b. in the performance of the Named Insured's ongoing operations.

But the coverage hereby granted to such additional insureds does not apply to **bodily injury**, **property damage** or **personal and advertising injury** arising out of the rendering of or failure to render any professional services by, on behalf of, or for the **Named Insured**, including but not limited to:

CNA74858XX (1-15)
Page 2 of 18
CONTINENTAL CASUALTY COMPANY

Endorsement No: 8
Effective Date: 11/22/2022

Policy No: 6056712992

- the preparing, approving, or failing to prepare or approve, maps, shop drawings, opinions, reports, surveys, field orders, change orders or drawings and specifications; or
- 2. supervisory, inspection, architectural or engineering activities.

D. Lessor of Equipment

Any person or organization from whom a Named Insured leases equipment, but only with respect to liability for bodily injury, property damage or personal and advertising injury caused, in whole or in part, by the Named Insured's maintenance, operation or use of such equipment, provided that the occurrence giving rise to such bodily injury, property damage or the offense giving rise to such personal and advertising injury takes place prior to the termination of such lease.

E. Lessor of Land

Any person or organization from whom a Named Insured leases land but only with respect to liability for bodily injury, property damage or personal and advertising injury arising out of the ownership, maintenance or use of such land, provided that the occurrence giving rise to such bodily injury, property damage or the offense giving rise to such personal and advertising injury takes place prior to the termination of such lease. The coverage granted by this paragraph does not apply to structural alterations, new construction or demolition operations performed by, on behalf of, or for such additional insured.

F. Lessor of Premises

An owner or lessor of premises leased to the Named Insured, or such owner or lessor's real estate manager, but only with respect to liability for bodily injury, property damage or personal and advertising injury arising out of the ownership, maintenance or use of such part of the premises leased to the Named Insured, and provided that the occurrence giving rise to such bodily injury or property damage, or the offense giving rise to such personal and advertising injury, takes place prior to the termination of such lease. The coverage granted by this paragraph does not apply to structural alterations, new construction or demolition operations performed by, on behalf of, or for such additional insured.

G. Mortgagee, Assignee or Receiver

A mortgagee, assignee or receiver of premises but only with respect to such mortgagee, assignee or receiver's liability for bodily injury, property damage or personal and advertising injury arising out of the Named Insured's ownership, maintenance, or use of a premises by a Named Insured.

The coverage granted by this paragraph does not apply to structural alterations, new construction or demolition operations performed by, on behalf of, or for such additional insured.

H. State or Governmental Agency or Subdivision or Political Subdivisions – Permits

A state or governmental agency or subdivision or political subdivision that has issued a permit or authorization but only with respect to such state or governmental agency or subdivision or political subdivision's liability for bodily injury, property damage or personal and advertising injury arising out of:

- the following hazards in connection with premises a Named Insured owns, rents, or controls and to which this insurance applies:
 - a. the existence, maintenance, repair, construction, erection, or removal of advertising signs, awnings, canopies, cellar entrances, coal holes, driveways, manholes, marquees, hoistaway openings, sidewalk vaults, street banners, or decorations and similar exposures; or
 - b. the construction, erection, or removal of elevators; or
 - c. the ownership, maintenance or use of any elevators covered by this insurance; or

CNA74858XX (1-15)

CONTINENTAL CASUALTY COMPANY

Page 3 of 18 Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: 8



Architects, Engineers and Surveyors General Liability Extension Endorsement

2. the permitted or authorized operations performed by a Named Insured or on a Named Insured's behalf.

The coverage granted by this paragraph does not apply to:

- a. Bodily injury, property damage or personal and advertising injury arising out of operations performed for the state or governmental agency or subdivision or political subdivision; or
- b. Bodily injury or property damage included within the products-completed operations hazard.

With respect to this provision's requirement that additional insured status must be requested under a written contract or agreement, the Insurer will treat as a written contract any governmental permit that requires the **Named Insured** to add the governmental entity as an additional insured.

I. Trade Show Event Lessor

- 1. With respect to a Named Insured's participation in a trade show event as an exhibitor, presenter or displayer, any person or organization whom the Named Insured is required to include as an additional insured, but only with respect to such person or organization's liability for bodily injury, property damage or personal and advertising injury caused by:
 - a. the Named Insured's acts or omissions; or
 - b. the acts or omissions of those acting on the Named Insured's behalf,

in the performance of the Named Insured's ongoing operations at the trade show event premises during the trade show event.

2. The coverage granted by this paragraph does not apply to **bodily injury** or **property damage** included within the **products-completed operations hazard**.

2. ADDITIONAL INSURED - PRIMARY AND NON-CONTRIBUTORY TO ADDITIONAL INSURED'S INSURANCE

The Other Insurance Condition in the COMMERCIAL GENERAL LIABILITY CONDITIONS Section is amended to add the following paragraph:

If the **Named Insured** has agreed in writing in a contract or agreement that this insurance is primary and non-contributory relative to an additional insured's own insurance, then this insurance is primary, and the Insurer will not seek contribution from that other insurance. For the purpose of this Provision **2.**, the additional insured's own insurance means insurance on which the additional insured is a named insured. Otherwise, and notwithstanding anything to the contrary elsewhere in this Condition, the insurance provided to such person or organization is excess of any other insurance available to such person or organization.

3. ADDITIONAL INSURED - EXTENDED COVERAGE

When an additional insured is added by this or any other endorsement attached to this Coverage Part, WHO IS AN INSURED is amended to make the following natural persons Insureds.

If the additional insured is:

- a. An individual, then his or her spouse is an Insured;
- b. A partnership or joint venture, then its partners, members and their spouses are Insureds;
- c. A limited liability company, then its members and managers are Insureds; or
- d. An organization other than a partnership, joint venture or limited liability company, then its executive officers, directors and shareholders are **Insureds**:

CNA74858XX (1-15)
Page 4 of 18
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: 8





but only with respect to locations and operations covered by the additional insured endorsement's provisions, and only with respect to their respective roles within their organizations.

Please see the ESTATES, LEGAL REPRESENTATIVES, AND SPOUSES provision of this endorsement for additional coverage and restrictions applicable to spouses of natural person Insureds.

4. BOATS

Under COVERAGES, Coverage A - Bodily Injury And Property Damage Liability, the paragraph entitled Exclusions is amended to add the following additional exception to the exclusion entitled Aircraft, Auto or Watercraft:

This exclusion does not apply to:

Any watercraft owned by the Named Insured that is less than 30 feet long while being used in the course of the Named Insured's inspection or surveying work.

5. BODILY INJURY - EXPANDED DEFINITION

Under **DEFINITIONS**, the definition of **bodily injury** is deleted and replaced by the following:

Bodily injury means physical injury, sickness or disease sustained by a person, including death, humiliation, shock, mental anguish or mental injury sustained by that person at any time which results as a consequence of the physical injury, sickness or disease.

BROAD KNOWLEDGE OF OCCURRENCE/ NOTICE OF OCCURRENCE

Under CONDITIONS, the condition entitled Duties in The Event of Occurrence, Offense, Claim or Suit is amended to add the following provisions:

A. BROAD KNOWLEDGE OF OCCURRENCE

The Named Insured must give the Insurer or the Insurer's authorized representative notice of an occurrence, offense or claim only when the occurrence, offense or claim is known to a natural person Named Insured, to a partner, executive officer, manager or member of a Named Insured, or to an employee designated by any of the above to give such notice.

B. NOTICE OF OCCURRENCE

The Named Insured's rights under this Coverage Part will not be prejudiced if the Named Insured fails to give the Insurer notice of an occurrence, offense or claim and that failure is solely due to the Named Insured's reasonable belief that the bodily injury or property damage is not covered under this Coverage Part. However, the Named Insured shall give written notice of such occurrence, offense or claim to the Insurer as soon as the Named Insured is aware that this insurance may apply to such occurrence, offense or claim.

BROAD NAMED INSURED

WHO IS AN INSURED is amended to delete its Paragraph 3. in its entirety and replace it with the following:

- Pursuant to the limitations described in Paragraph 4. below, any organization in which a Named Insured has management control:
 - a. on the effective date of this Coverage Part; or
 - by reason of a Named Insured creating or acquiring the organization during the policy period,

qualifies as a Named Insured, provided that there is no other similar liability insurance, whether primary, contributory, excess, contingent or otherwise, which provides coverage to such organization, or which would have

CNA74858XX (1-15)

Page 5 of 18

CONTINENTAL CASUALTY COMPANY

Insured Name: MOFFATT & NICHOL

Policy No: 6056712992

Endorsement No:

Effective Date: 11/22/2022

8



Architects, Engineers and Surveyors General Liability Extension Endorsement

provided coverage but for the exhaustion of its limit, and without regard to whether its coverage is broader or narrower than that provided by this insurance.

But this BROAD NAMED INSURED provision does not apply to:

- (a) any partnership, limited liability company or joint venture; or
- (b) any organization for which coverage is excluded by another endorsement attached to this Coverage Part.

For the purpose of this provision, management control means:

- A. owning interests representing more than 50% of the voting, appointment or designation power for the selection of a majority of the Board of Directors of a corporation; or
- B. having the right, pursuant to a written trust agreement, to protect, control the use of, encumber or transfer or sell property held by a trust.
- **4.** With respect to organizations which qualify as **Named Insureds** by virtue of Paragraph **3.** above, this insurance does not apply to:
 - a. bodily injury or property damage that first occurred prior to the date of management control, or that first occurs after management control ceases; nor
 - b. personal or advertising injury caused by an offense that first occurred prior to the date of management control or that first occurs after management control ceases.
- 5. The insurance provided by this Coverage Part applies to Named Insureds when trading under their own names or under such other trading names or doing-business-as names (dba) as any Named Insured should choose to employ.

8. CONTRACTUAL LIABILITY - RAILROADS

With respect to operations performed within 50 feet of railroad property, the definition of **insured contract** is replaced by the following:

Insured Contract means:

- a. A contract for a lease of premises. However, that portion of the contract for a lease of premises that indemnifies any person or organization for damage by fire to premises while rented to a Named Insured or temporarily occupied by a Named Insured you with permission of the owner is not an insured contract;
- b. A sidetrack agreement;
- c. Any easement or license agreement;
- **d.** An obligation, as required by ordinance, to indemnify a municipality, except in connection with work for a municipality;
- e. An elevator maintenance agreement;
- f. That part of any other contract or agreement pertaining to the Named Insured's business (including an indemnification of a municipality in connection with work performed for a municipality) under which the Named Insured assumes the tort liability of another party to pay for bodily injury or property damage to a third person or organization. Tort liability means a liability that would be imposed by law in the absence of any contract or agreement.

Paragraph f. does not include that part of any contract or agreement:

CNA74858XX (1-15)
Page 6 of 18
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Policy No: 6056712992



- (1) That indemnifies an architect, engineer or surveyor for injury or damage arising out of:
 - (a) Preparing, approving or failing to prepare or approve maps, shop drawings, opinions, reports, surveys, field orders, change orders or drawings and specifications; or
 - (b) Giving directions or instructions, or failing to give them, if that is the primary cause of the injury or damage;
- (2) Under which the Insured, if an architect, engineer or surveyor, assumes liability for an injury or damage arising out of the insured's rendering or failure to render professional services, including those listed in (1) above and supervisory, inspection, architectural or engineering activities.

9. ESTATES, LEGAL REPRESENTATIVES, AND SPOUSES

The estates, executors, heirs, legal representatives, administrators, trustees, beneficiaries and spouses of any natural person Insured or living trust shall also be insured under this policy; provided, however, coverage is afforded to such estates, executors, heirs, legal representatives, administrators, trustees, beneficiaries and spouses only for claims arising solely out of their capacity or status as such and, in the case of a spouse, where such claim seeks damages from marital community property, jointly held property or property transferred from such natural person Insured to such spouse. No coverage is provided for any act, error or omission of an estate, heir, legal representative, or spouse outside the scope of such person's capacity or status as such, provided, however, that the spouse of a natural person Named Insured, and the spouses of members or partners of joint venture or partnership Named Insureds are Insureds with respect to such spouses' acts, errors or omissions in the conduct of the Named Insured's business.

10. EXPECTED OR INTENDED INJURY - EXCEPTION FOR REASONABLE FORCE

Under COVERAGES, Coverage A - Bodily Injury And Property Damage Liability, the paragraph entitled Exclusions is amended to delete the exclusion entitled Expected or Intended Injury and replace it with the following:

This insurance does not apply to:

Expected or Intended Injury

Bodily injury or property damage expected or intended from the standpoint of the Insured. This exclusion does not apply to **bodily injury** or **property damage** resulting from the use of reasonable force to protect persons or property.

11. GENERAL AGGREGATE LIMITS OF INSURANCE - PER LOCATION

- A. A separate Location General Aggregate Limit, equal to the amount of the General Aggregate Limit, is the most the Insurer will pay for the sum of:
 - 1. All damages under Coverage A, except damages because of bodily injury or property damage included in the products-completed operations hazard; and
 - 2. All medical expenses under Coverage C.

that arise from occurrences or accidents which can be attributed solely to ongoing operations at that location. Such payments shall not reduce the General Aggregate Limit shown in the Declarations, nor the Location General Aggregate Limit of any other location.

B. All:

CNA74858XX (1-15)

1. Damages under Coverage B, regardless of the number of locations involved;

50020009260567129921752

Policy No: 6056712992 **Endorsement No:** 8 Effective Date: 11/22/2022

Page 7 of 18 CONTINENTAL CASUALTY COMPANY Insured Name: MOFFATT & NICHOL



Architects, Engineers and Surveyors General Liability Extension Endorsement

- 2. Damages under Coverage A, caused by occurrences which cannot be attributed solely to ongoing operations at a single location, except damages because of bodily injury or property damage included in the products-completed operations hazard; and
- 3. Medical expenses under Coverage C caused by accidents which cannot be attributed solely to ongoing operations at a single location,

will reduce the General Aggregate Limit shown in the Declarations.

- C. For the purpose of this GENERAL AGGREGATE LIMITS OF INSURANCE PER LOCATION Provision, "location" means:
 - 1. a premises the Named Insured owns or rents; or
 - 2. a premises not owned or rented by any Named Insured at which the Named Insured is performing operations pursuant to a contract or written agreement. If operations at such a location have been discontinued and then restarted, or if the authorized parties deviate from plans, blueprints, designs, specifications or timetables, the location will still be deemed to be the same location.

For the purpose of determining the applicable aggregate limit of insurance, premises involving the same or connecting lots, or premises whose connection is interrupted only by a street, roadway, waterway or right-of-way of a railroad shall be considered a single location.

- D. The limits shown in the Declarations for Each Occurrence, for Damage To Premises Rented To You and for Medical Expense continue to apply, but will be subject to either the Location General Aggregate Limit or the General Aggregate Limit, depending on whether the occurrence can be attributed solely to ongoing operations at a particular location.
- E. When coverage for liability arising out of the products-completed operations hazard is provided, any payments for damages because of bodily injury or property damage included in the products-completed operations hazard, regardless of the number of locations involved, will reduce the Products-Completed Operations Aggregate Limit shown in the Declarations.
- F. The provisions of LIMITS OF INSURANCE not otherwise modified by this GENERAL AGGREGATE LIMITS OF **INSURANCE - PER LOCATION** Provision shall continue to apply as stipulated.

12. IN REM ACTIONS

A quasi in rem action against any vessel owned or operated by or for the Named Insured, or chartered by or for the Named Insured, will be treated in the same manner as though the action were in personam against the Named Insured.

13. INCIDENTAL HEALTH CARE MALPRACTICE COVERAGE

Solely with respect to **bodily injury** that arises out of a **health care incident**:

- A. Under COVERAGES, Coverage A Bodily Injury And Property Damage Liability, the Insuring Agreement is amended to replace Paragraphs 1.b.(1) and 1.b.(2) with the following:
 - b. This insurance applies to bodily injury provided that the professional health care services are incidental to the Named Insured's primary business purpose, and only if:
 - (1) such bodily injury is caused by an occurrence that takes place in the coverage territory.
 - (2) the bodily injury first occurs during the policy period. All bodily injury arising from an occurrence will be deemed to have occurred at the time of the first act, error, or omission that is part of the occurrence; and

CNA74858XX (1-15)

Page 8 of 18

CONTINENTAL CASUALTY COMPANY

Insured Name: MOFFATT & NICHOL

Policy No: 6056712992

Endorsement No:

8



- B. Under COVERAGES, Coverage A Bodily Injury And Property Damage Liability, the paragraph entitled Exclusions is amended to:
 - add the following to the Employers Liability exclusion:

This exclusion applies only if the **bodily injury** arising from a **health care incident** is covered by other liability insurance available to the Insured (or which would have been available but for exhaustion of its limits).

ii. delete the exclusion entitled Contractual Liability and replace it with the following:

This insurance does not apply to:

Contractual Liability

the Insured's actual or alleged liability under any oral or written contract or agreement, including but not limited to express warranties or guarantees.

iii. to add the following additional exclusions:

This insurance does not apply to:

Discrimination

any actual or alleged discrimination, humiliation or harassment, including but not be limited to claims based on an individual's race, creed, color, age, gender, national origin, religion, disability, marital status or sexual orientation.

Dishonesty or Crime

Any actual or alleged dishonest, criminal or malicious act, error or omission.

Medicare/Medicaid Fraud

any actual or alleged violation of law with respect to Medicare, Medicaid, Tricare or any similar federal, state or local governmental program.

Services Excluded by Endorsement

Any health care incident for which coverage is excluded by endorsement.

- C. **DEFINITIONS** is amended to:
 - add the following definitions:

Health care incident means an act, error or omission by the Named Insured's employees or volunteer workers in the rendering of:

- professional health care services on behalf of the Named Insured or
- Good Samaritan services rendered in an emergency and for which no payment is demanded or received.

Professional health care services means any health care services or the related furnishing of food, beverages, medical supplies or appliances by the following providers in their capacity as such but solely to the extent they are duly licensed as required:

a. Physician;

50020009260567129921753

CNA74858XX (1-15) Page 9 of 18 CONTINENTAL CASUALTY COMPANY Insured Name: MOFFATT & NICHOL

Endorsement No: 8 Effective Date: 11/22/2022

6056712992

Policy No:

Copyright CNA All Rights Reserved. Includes copyrighted material of Insurance Services Office, Inc., with its permission.



Policy No: 6056712992

Endorsement No:

Architects, Engineers and Surveyors General Liability Extension Endorsement

- b. Nurse;
- c. Nurse practitioner;
- d. Emergency medical technician;
- e. Paramedic;
- f. Dentist;
- g. Physical therapist;
- h. Psychologist;
- Speech therapist;
- Other allied health professional; or

Professional health care services does not include any services rendered in connection with human clinical trials or product testing.

ii. delete the definition of occurrence and replace it with the following:

Occurrence means a health care incident. All acts, errors or omissions that are logically connected by any common fact, circumstance, situation, transaction, event, advice or decision will be considered to constitute a single occurrence;

- iii. amend the definition of Insured to:
 - a. add the following:

the Named Insured's employees are Insureds with respect to:

- (1) bodily injury to a co-employee while in the course of the co-employee's employment by the Named Insured or while performing duties related to the conduct of the Named Insured's business; and
- (2) bodily injury to a volunteer worker while performing duties related to the conduct of the Named Insured's business;

when such bodily injury arises out of a health care incident.

the Named Insured's volunteer workers are Insureds with respect to:

- (1) bodily injury to a co-volunteer worker while performing duties related to the conduct of the Named Insured's business; and
- (2) bodily injury to an employee while in the course of the employee's employment by the Named Insured or while performing duties related to the conduct of the Named Insured's business;

when such bodily injury arises out of a health care incident.

- b. delete Subparagraphs (a), (b), (c) and (d) of Paragraph 2.a.(1) of WHO IS AN INSURED.
- D. The Other Insurance condition is amended to delete Paragraph b.(1) in its entirety and replace it with the following:

Other Insurance

CNA74858XX (1-15)
Page 10 of 18
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL



b. Excess Insurance

(1) To the extent this insurance applies, it is excess over any other insurance, self insurance or risk transfer instrument, whether primary, excess, contingent or on any other basis, except for insurance purchased specifically by the Named Insured to be excess of this coverage.

14. JOINT VENTURES / PARTNERSHIP / LIMITED LIABILITY COMPANIES

A. Past Joint Ventures, Partnerships, Limited Liability Companies

The following is added to WHO IS AN INSURED:

If the Named Insured was a joint venturer, partner, or member of a limited liability company and such joint venture, partnership or limited liability company terminated prior to or during the **policy period**, such Named Insured is an Insured with respect to its interest in such joint venture, partnership or limited liability company but only to the extent that:

- a. any offense giving rise to personal and advertising injury occurred prior to such termination date, and the personal and advertising injury arising out of such offense, first occurred after such termination date;
- b. the bodily injury or property damage first occurred after such termination date; and
- c. there is no other valid and collectible insurance purchased specifically to insure the partnership, joint venture or limited liability company.

If the joint venture, partnership or limited liability company is or was insured under a consolidated (wrap-up) insurance program, then such insurance will always be considered valid and collectible for the purpose of paragraph c. above. But this provision will not serve to exclude bodily injury, property damage or personal and advertising injury that would otherwise be covered under the Architects, Engineers And Surveyors General Liability Extension Endorsement provision entitled WRAP-UP EXTENSION: OCIP, CCIP, OR CONSOLIDATED (WRAP-UP) INSURANCE PROGRAMS. Please see that provision for the definition of consolidated (wrap-up) insurance program.

B. Participation In Current Professional Joint Ventures

The following is added to WHO IS AN INSURED:

The **Named Insured** is also an **Insured** for participation in a current joint venture that is not named on the Declarations, but only if such joint venture meets all of the following criteria:

- Each and every one of the Named Insured's co-venturers are architectural, engineering or surveying firms only; and
- b. There is no other valid and collectible insurance purchased specifically to insure the joint venture.

However, the **Named Insured** is an **Insured** only for the conduct of such **Named Insured's** business within such a joint venture. The **Named Insured** is not insured for liability arising out of the acts or omissions of other coventurers, nor of their partners, members or employees.

C. WHO IS AN INSURED is amended to delete its last paragraph and replace it with the following:

Except as provided under this Architects, Engineers And Surveyors General Liability Extension Endorsement or by the attachment of another endorsement (if any), no person or organization is an Insured with respect to the conduct of any current or past partnership, joint venture or limited liability company that is not shown as a Named Insured in the Declarations.

50020009260567129921754



CNA74858XX (1-15)
Page 11 of 18
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: 8 Effective Date: 11/22/2022

Copyright CNA All Rights Reserved. Includes copyrighted material of Insurance Services Office, Inc., with its permission.



Architects, Engineers and Surveyors General Liability Extension Endorsement

- 15. LEGAL LIABILITY DAMAGE TO PREMISES / ALIENATED PREMISES / PROPERTY IN THE NAMED INSURED'S CARE, CUSTODY OR CONTROL
 - A. Under COVERAGES, Coverage A Bodily Injury and Property Damage Liability, the paragraph entitled Exclusions is amended to delete exclusion j. Damage to Property in its entirety and replace it with the following:

This insurance does not apply to:

j. Damage to Property

Property damage to:

- (1) Property the Named Insured owns, rents, or occupies, including any costs or expenses incurred by you, or any other person, organization or entity, for repair, replacement, enhancement, restoration or maintenance of such property for any reason, including prevention of injury to a person or damage to another's property;
- (2) Premises the Named Insured sells, gives away or abandons, if the property damage arises out of any part of those premises;
- (3) Property loaned to the Named Insured;
- (4) Personal property in the care, custody or control of the Insured;
- (5) That particular part of real property on which the **Named Insured** or any contractors or subcontractors working directly or indirectly on the **Named Insured's** behalf are performing operations, if the **property damage** arises out of those operations; or
- (6) That particular part of any property that must be restored, repaired or replaced because your work was incorrectly performed on it.

Paragraphs (1), (3) and (4) of this exclusion do not apply to **property damage** (other than damage by fire) to premises rented to the **Named Insured** or temporarily occupied by the **Named Insured** with the permission of the owner, nor to the contents of premises rented to the **Named Insured** for a period of 7 or fewer consecutive days. A separate limit of insurance applies to Damage To Premises Rented To You as described in **LIMITS OF INSURANCE**.

Paragraph (2) of this exclusion does not apply if the premises are your work.

Paragraphs (3), (4), (5) and (6) of this exclusion do not apply to liability assumed under a sidetrack agreement.

Paragraph (6) of this exclusion does not apply to property damage included in the products-completed operations hazard.

Paragraphs (3) and (4) of this exclusion do not apply to property damage to:

- i. tools, or equipment the Named Insured borrows from others, nor
- ii. other personal property of others in the Named Insured's care, custody or control while being used in the Named Insured's operations away from any Named Insured's premises.

However, the coverage granted by this exception to Paragraphs (3) and (4) does not apply to:

- a. property at a job site awaiting or during such property's installation, fabrication, or erection;
- b. property that is mobile equipment leased by an Insured;

CNA74858XX (1-15) Policy No: 6056712992
Page 12 of 18 Endorsement No: 8

CONTINENTAL CASUALTY COMPANY Insured Name: MOFFATT & NICHOL



- c. property that is an auto, aircraft or watercraft;
- d. property in transit; or
- e. any portion of property damage for which the Insured has available other valid and collectible insurance, or would have such insurance but for exhaustion of its limits, or but for application of one of its exclusions.

A separate limit of insurance and deductible apply to such property of others. See LIMITS OF INSURANCE as amended below.

B. Under COVERAGES, Coverage A - Bodily Injury and Property Damage Liability, the paragraph entitled **Exclusions** is amended to delete its last paragraph and replace it with the following:

Exclusions c. through n. do not apply to damage by fire to premises while rented to a Named Insured or temporarily occupied by a Named Insured with permission of the owner, nor to damage to the contents of premises rented to a Named Insured for a period of 7 or fewer consecutive days.

A separate limit of insurance applies to this coverage as described in LIMITS OF INSURANCE.

C. The following paragraph is added to **LIMITS OF INSURANCE**:

Subject to 5. above, \$25,000 is the most the Insurer will pay under Coverage A for damages arising out of any one occurrence because of the sum of all property damage to borrowed tools or equipment, and to other personal property of others in the Named Insured's care, custody or control, while being used in the Named Insured's operations away from any Named Insured's premises. The Insurer's obligation to pay such property damage does not apply until the amount of such property damage exceeds \$1,000. The Insurer has the right but not the duty to pay any portion of this \$1,000 in order to effect settlement. If the Insurer exercises that right, the Named Insured will promptly reimburse the Insurer for any such amount.

- D. Paragraph 6., Damage To Premises Rented To You Limit, of LIMITS OF INSURANCE is deleted and replaced by the following:
 - 6. Subject to Paragraph 5. above, (the Each Occurrence Limit), the Damage To Premises Rented To You Limit is the most the Insurer will pay under Coverage A for damages because of property damage to any one premises while rented to the Named Insured or temporarily occupied by the Named Insured with the permission of the owner, including contents of such premises rented to the Named Insured for a period of 7 or fewer consecutive days. The Damage To Premises Rented To You Limit is the greater of:
 - a. \$500,000; or
 - b. The Damage To Premises Rented To You Limit shown in the Declarations.
- E. Paragraph 4.b.(1)(a)(ii) of the Other Insurance Condition is deleted and replaced by the following:
 - (ii) That is property insurance for premises rented to the Named Insured, for premises temporarily occupied by the Named Insured with the permission of the owner; or for personal property of others in the Named insured's care, custody or control;

16. LIQUOR LIABILITY

Under COVERAGES, Coverage A - Bodily Injury and Property Damage Liability, the paragraph entitled **Exclusions** is amended to delete the exclusion entitled Liquor Liability.

This LIQUOR LIABILITY Provision does not apply to any person or organization who otherwise qualifies as an additional insured on this Coverage Part.

CNA74858XX (1-15) Page 13 of 18 CONTINENTAL CASUALTY COMPANY Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 **Endorsement No:** 8 Effective Date: 11/22/2022



Architects, Engineers and Surveyors General Liability Extension Endorsement

17. MEDICAL PAYMENTS

- A. LIMITS OF INSURANCE is amended to delete Paragraph 7. (the Medical Expense Limit) and replace it with the following:
 - 7. Subject to Paragraph 5. above (the Each Occurrence Limit), the Medical Expense Limit is the most the Insurer will pay under Coverage C for all medical expenses because of bodily injury sustained by any one person. The Medical Expense Limit is the greater of:
 - (1) \$15,000 unless a different amount is shown here: \$N,NNN,NNN,NNN; or
 - (2) the amount shown in the Declarations for Medical Expense Limit.
- B. Under COVERAGES, the Insuring Agreement of Coverage C Medical Payments is amended to replace Paragraph 1.a.(3)(b) with the following:
 - (b) The expenses are incurred and reported to the Insurer within three years of the date of the accident; and

18. NON-OWNED AIRCRAFT

Under COVERAGES, Coverage A – Bodily Injury and Property Damage Liability, the paragraph entitled Exclusions is amended as follows:

The exclusion entitled Aircraft, Auto or Watercraft is amended to add the following:

This exclusion does not apply to an aircraft not owned by any Named Insured, provided that:

- 1. the pilot in command holds a currently effective certificate issued by the duly constituted authority of the United States of America or Canada, designating that person as a commercial or airline transport pilot;
- 2. the aircraft is rented with a trained, paid crew to the Named Insured; and
- 3. the aircraft is not being used to carry persons or property for a charge.

19. NON-OWNED WATERCRAFT

Under COVERAGES, Coverage A – Bodily Injury and Property Damage Liability, the paragraph entitled Exclusions is amended to delete subparagraph (2) of the exclusion entitled Aircraft, Auto or Watercraft, and replace it with the following.

This exclusion does not apply to:

- (2) a watercraft that is not owned by any Named Insured, provided the watercraft is:
 - (a) less than 75 feet long; and
 - (b) not being used to carry persons or property for a charge.

20. PERSONAL AND ADVERTISING INJURY -DISCRIMINATION OR HUMILIATION

- A. Under **DEFINITIONS**, the definition of **personal and advertising injury** is amended to add the following tort:
 - Discrimination or humiliation that results in injury to the feelings or reputation of a natural person.
- B. Under COVERAGES, Coverage B Personal and Advertising Injury Liability, the paragraph entitled Exclusions is amended to:
 - 1. delete the Exclusion entitled Knowing Violation Of Rights Of Another and replace it with the following:

CNA74858XX (1-15)
Page 14 of 18
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: 8 Effective Date: 11/22/2022

Copyright CNA All Rights Reserved. Includes copyrighted material of Insurance Services Office, Inc., with its permission.



This insurance does not apply to:

Knowing Violation of Rights of Another

Personal and advertising injury caused by or at the direction of the **Insured** with the knowledge that the act would violate the rights of another and would inflict **personal and advertising injury**. This exclusion shall not apply to discrimination or humiliation that results in injury to the feelings or reputation of a natural person, but only if such discrimination or humiliation is not done intentionally by or at the direction of:

- (a) the Named Insured; or
- (b) any executive officer, director, stockholder, partner, member or manager (if the Named Insured is a limited liability company) of the Named Insured.
- 2. add the following exclusions:

This insurance does not apply to:

Employment Related Discrimination

discrimination or humiliation directly or indirectly related to the employment, prospective employment, past employment or termination of employment of any person by any **Insured**.

Premises Related Discrimination

discrimination or humiliation arising out of the sale, rental, lease or sub-lease or prospective sale, rental, lease or sub-lease of any room, dwelling or premises by or at the direction of any **Insured**.

Notwithstanding the above, there is no coverage for fines or penalties levied or imposed by a governmental entity because of discrimination.

The coverage provided by this **PERSONAL AND ADVERTISING INJURY –DISCRIMINATION OR HUMILIATION** Provision does not apply to any person or organization whose status as an **Insured** derives solely from

Provision 1. ADDITIONAL INSURED of this endorsement; or

attachment of an additional insured endorsement to this Coverage Part.

21. PERSONAL AND ADVERTISING INJURY - CONTRACTUAL LIABILITY

- A. Under COVERAGES, Coverage B —Personal and Advertising Injury Liability, the paragraph entitled Exclusions is amended to delete the exclusion entitled Contractual Liability.
- B. Solely for the purpose of the coverage provided by this PERSONAL AND ADVERTISING INJURY LIMITED CONTRACTUAL LIABILITY provision, the following changes are made to the section entitled SUPPLEMENTARY PAYMENTS COVERAGES A AND B:
 - 1. Paragraph 2.d. is replaced by the following:
 - d. The allegations in the suit and the information the Insurer knows about the offense alleged in such suit are such that no conflict appears to exist between the interests of the Insured and the interests of the indemnitee;
 - 2. The first unnumbered paragraph beneath Paragraph 2.f.(2)(b) is deleted and replaced by the following:

So long as the above conditions are met, attorney's fees incurred by the Insurer in the defense of that indemnitee, necessary litigation expenses incurred by the Insurer, and necessary litigation expenses incurred

50020009260567129921756

Policy No:

Endorsement No:

6056712992

R



Effective Date: 11/22/2022

Architects, Engineers and Surveyors General Liability Extension Endorsement

by the indemnitee at the Insurer's request will be paid as **defense costs**. Such payments will not be deemed to be **damages** for **personal and advertising injury** and will not reduce the limits of insurance.

C. This PERSONAL AND ADVERTISING INJURY - LIMITED CONTRACTUAL LIABILITY Provision does not apply if Coverage B -Personal and Advertising Injury Liability is excluded by another endorsement attached to this Coverage Part.

This PERSONAL AND ADVERTISING INJURY - CONTRACTUAL LIABILITY Provision does not apply to any person or organization who otherwise qualifies as an additional insured on this Coverage Part.

22. PROPERTY DAMAGE - ELEVATORS

- A. Under COVERAGES, Coverage A Bodily Injury and Property Damage Liability, the paragraph entitled Exclusions is amended such that the Damage to Your Product Exclusion and subparagraphs (3), (4) and (6) of the Damage to Property Exclusion do not apply to property damage that results from the use of elevators.
- B. Solely for the purpose of the coverage provided by this **PROPERTY DAMAGE ELEVATORS** Provision, the **Other Insurance** conditions is amended to add the following paragraph:

This insurance is excess over any of the other insurance, whether primary, excess, contingent or on any other basis that is Property insurance covering property of others damaged from the use of elevators.

23. RETIRED PARTNERS, MEMBERS, DIRECTORS AND EMPLOYEES

WHO IS INSURED is amended to include as Insureds natural persons who are retired partners, members, directors or employees, but only for bodily injury, property damage or personal and advertising injury that results from services performed for the Named Insured under the Named Insured's direct supervision. All limitations that apply to employees and volunteer workers also apply to anyone qualifying as an Insured under this Provision.

24. SUPPLEMENTARY PAYMENTS

The section entitled SUPPLEMENTARY PAYMENTS - COVERAGES A AND B is amended as follows:

- A. Paragraph 1.b. is amended to delete the \$250 limit shown for the cost of bail bonds and replace it with a \$5,000. limit: and
- B. Paragraph 1.d. is amended to delete the limit of \$250 shown for daily loss of earnings and replace it with a \$1,000. limit.

25. UNINTENTIONAL FAILURE TO DISCLOSE HAZARDS

If the Named Insured unintentionally fails to disclose all existing hazards at the inception date of the Named Insured's Coverage Part, the Insurer will not deny coverage under this Coverage Part because of such failure.

26. WAIVER OF SUBROGATION - BLANKET

Under CONDITIONS, the condition entitled Transfer Of Rights Of Recovery Against Others To Us is amended to add the following:

The Insurer waives any right of recovery the Insurer may have against any person or organization because of payments the Insurer makes for injury or damage arising out of:

- 1. the Named Insured's ongoing operations; or
- 2. your work included in the products-completed operations hazard.

However, this waiver applies only when the **Named Insured** has agreed in writing to waive such rights of recovery in a written contract or written agreement, and only if such contract or agreement:

CNA74858XX (1-15) Policy No: 6056712992
Page 16 of 18 Endorsement No: 8

Page 16 of 18
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL



- 1. is in effect or becomes effective during the term of this Coverage Part; and
- 2. was executed prior to the bodily injury, property damage or personal and advertising injury giving rise to the claim.

27. WRAP-UP EXTENSION: OCIP. CCIP. OR CONSOLIDATED (WRAP-UP) INSURANCE PROGRAMS

Note: The following provision does not apply to any public construction project in the state of Oklahoma, nor to any construction project in the state of Alaska, that is not permitted to be insured under a consolidated (wrap-up) insurance program by applicable state statute or regulation.

If the endorsement EXCLUSION - CONSTRUCTION WRAP-UP is attached to this policy, or another exclusionary endorsement pertaining to Owner Controlled Insurance Programs (O.C.I.P.) or Contractor Controlled Insurance Programs (C.C.I.P.) is attached, then the following changes apply:

A. The following wording is added to the above-referenced endorsement:

With respect to a consolidated (wrap-up) insurance program project in which the Named Insured is or was involved, this exclusion does not apply to those sums the Named Insured become legally obligated to pay as damages because of:

- 1. Bodily injury, property damage, or personal or advertising injury that occurs during the Named Insured's ongoing operations at the project, or during such operations of anyone acting on the Named Insured's behalf; nor
- 2. Bodily injury or property damage included within the products-completed operations hazard that arises out of those portions of the project that are not residential structures.
- B. Condition 4. Other Insurance is amended to add the following subparagraph 4.b.(1)(c):

This insurance is excess over:

- (c) Any of the other insurance whether primary, excess, contingent or any other basis that is insurance available to the Named Insured as a result of the Named Insured being a participant in a consolidated (wrap-up) insurance program, but only as respects the Named Insured's involvement in that consolidated (wrap-up) insurance program.
- **C. DEFINITIONS** is amended to add the following definitions:

Consolidated (wrap-up) insurance program means a construction, erection or demolition project for which the prime contractor/project manager or owner of the construction project has secured general liability insurance covering some or all of the contractors or subcontractors involved in the project, such as an Owner Controlled Insurance Program (O.C.I.P.) or Contractor Controlled Insurance Program (C.C.I.P.).

Residential structure means any structure where 30% or more of the square foot area is used or is intended to be used for human residency, including but not limited to:

- single or multifamily housing, apartments, condominiums, townhouses, co-operatives or planned unit developments; and
- the common areas and structures appurtenant to the structures in paragraph 1. (including pools, hot tubs, detached garages, quest houses or any similar structures).

However, when there is no individual ownership of units, residential structure does not include military housing, college/university housing or dormitories, long term care facilities, hotels or motels. Residential structure also does not include hospitals or prisons.

CNA74858XX (1-15)

Page 17 of 18

CONTINENTAL CASUALTY COMPANY Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 8

Endorsement No:



Architects, Engineers and Surveyors General Liability Extension Endorsement

This WRAP-UP EXTENSION: OCIP, CCIP, OR CONSOLIDATED (WRAP-UP) INSURANCE PROGRAMS Provision does not apply to any person or organization who otherwise qualifies as an additional insured on this Coverage Part.

All other terms and conditions of the Policy remain unchanged.

This endorsement, which forms a part of and is for attachment to the Policy issued by the designated Insurers, takes effect on the effective date of said Policy at the hour stated in said Policy, unless another effective date is shown below, and expires concurrently with said Policy.

Insured Name: MOFFATT & NICHOL



General Aggregate Limit - Per Project Endorsement

This endorsement modifies insurance provided under the following:

COMMERCIAL GENERAL LIABILITY COVERAGE PART

It is understood and agreed as follows:

- For each single construction or service project away from premises the Named Insured owns or rents, a separate Project General Aggregate Limit, equal to the amount of the General Aggregate Limit shown in the Declarations, is the most the Insurer will pay for the sum of:
 - A. all damages under Coverage A, except damages because of bodily injury or property damage included in the products-completed operations hazard; and
 - B. all medical expenses under Coverage C;

that arise from occurrences or accidents which can be attributed solely to ongoing operations at that project. Such payments shall not reduce the General Aggregate Limit shown in the Declarations, nor the Project General Aggregate Limit applicable to any other project.

- II. All:
 - A. damages under Coverage B, regardless of the number of locations or projects involved;
 - B. damages under Coverage A, caused by occurrences which cannot be attributed solely to ongoing operations at a single project, except damages because of bodily injury or property damage included in the products-completed operations hazard; and
 - C. medical expenses under Coverage C, caused by accidents which cannot be attributed solely to ongoing operations at a single project,

will reduce the General Aggregate Limit shown in the Declarations.

- III. The limits shown in the Declarations for Each Occurrence, for Damage To Premises Rented To You and for Medical Expense continue to apply, but will be subject to either the Project General Aggregate Limit or the General Aggregate Limit shown in the Declarations, depending on whether the occurrence can be attributed solely to ongoing operations at a particular project.
- IV. When coverage for liability arising out of the products-completed operations hazard is provided, any payments for damages because of bodily injury or property damage included in the products-completed operations hazard will reduce the Products-Completed Operations Aggregate Limit shown in the Declarations, regardless of the number of projects involved.
- V. If a single construction or service project away from premises owned by or rented to the **Named Insured** has been abandoned and then restarted, or if the authorized contracting parties deviate from plans, blueprints, designs, specifications or timetables, such project will still be deemed to be the same project.
- VI. The provisions of LIMITS OF INSURANCE not otherwise modified by this endorsement shall continue to apply as stipulated.

All other terms and conditions of the Policy remain unchanged.

This endorsement, which forms a part of and is for attachment to the Policy issued by the designated Insurers, takes effect on the effective date of said Policy at the hour stated in said Policy, unless another effective date is shown below, and expires concurrently with said Policy.

20020009260567129925375



CNA75061XX (1-15)
Page 1 of 1
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Cancellation / Nonrenewal - California

Wherever used in this endorsement: 1) Insurer means "we", "us", "our" or the "Company" as those terms may be defined in the policy; and 2) Named Insured means the first person or entity named on the declarations page; and 3) "Insureds" means all persons or entities afforded coverage under the policy.

Any cancellation, nonrenewal or termination provisions in the policy are deleted in their entirety and replaced with the following:

CANCELLATION AND NONRENEWAL

A. CANCELLATION

- 1. The Named Insured may cancel the policy at any time. To do so, the Named Insured must return the policy to the Insurer or any of its authorized representatives, indicating the effective date of cancellation; or provide a written notice to the Insurer, stating when the cancellation is to be effective.
- 2. If the policy has been in effect for less than sixty (60) days and is not a renewal the Insurer may cancel the policy for any reason by mailing or delivering written notice to the Named Insured, at the last mailing address known to the Insurer, and the producer of record. The notice of cancellation will be provided at least sixty (60) days prior to the effective date of cancellation except that in the case of cancellation for nonpayment of premiums the notice will be given no less than ten (10) days prior to the effective date of the cancellation.
- 3. If the policy has been in effect for more than sixty (60) days or if it is a renewal, effective immediately, the Insurer may not cancel the policy unless such cancellation is based on one or more of the following reasons:
 - a. Nonpayment of premium, including payment due on a prior policy issued by the Insurer and due during the current policy term covering the same risks.
 - **b.** A judgment by a court or an administrative tribunal that the Named Insured has violated any law of this state or of the United States having as one of its necessary elements an act which materially increases any of the risks insured against.
 - c. Discovery of fraud or material misrepresentation by either of the following:
 - (1) The Named Insured or Insured(s) or a representative of same in obtaining the insurance; or
 - (2) The Named Insured or his or her representative in pursuing a claim under the policy.
 - d. Discovery of willful or grossly negligent acts or omissions, or of any violations of state laws or regulations establishing safety standards, by the Named Insured or Insured(s) or a representative of same, which materially increase any of the risks insured against.
 - e. Failure by the Named Insured or Insured(s) or a representative of same to implement reasonable loss control requirements which were agreed to by the Named Insured as a condition of policy issuance or which were conditions precedent to the use by the Insurer of a particular rate or rating plan, if the failure materially increases any of the risks insured against.
 - **f.** A determination by the commissioner that the loss of, or changes in, the Insurer's reinsurance covering all or part of the risk would threaten the financial integrity or solvency of the Insurer.
 - **g.** A determination by the commissioner that a continuation of the policy coverage would place the Insurer in violation of the laws of this state or the state of its domicile or that the continuation of coverage would threaten the solvency of the Insurer.
 - h. A change by the Named Insured or Insured(s) or a representative of same in the activities or property of the commercial or industrial enterprise which results in a material added risk, a materially increased risk or a materially changed risk, unless the added, increased, or changed risk is included in the policy.

A notice of cancellation will be in writing and will be delivered or mailed to the Named Insured, at the last mailing address known to the Insurer, and the producer of record at least sixty (60) days prior to the effective date of cancellation. Where cancellation is for nonpayment of premium, notice shall be given no less than ten (10) days prior to the effective date of cancellation.

40020009260567129923567



CNA62814CA (7-20)
Page 1 of 4
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: Effective Date: 11/22/2022



Cancellation / Nonrenewal – California

- 4. The notice will state the actual reason for the cancellation.
- 5. Notice of cancellation will state the effective date of cancellation. The policy period will end on that date.
- 6. If notice is mailed, proof of mailing will be sufficient proof of notice.

B. PREMIUM REFUND

If this policy is cancelled, the Insurer will send the Named Insured any premium refund due. If the Insurer cancels the refund will be pro rata. If the Named Insured cancels, the refund may be less than pro rata. The cancellation will be effective even if the Insurer has not made or offered a refund.

C. NONRENEWAL

- 1. The Insurer can non-renew the policy by giving written notice to the Named Insured, at the last mailing address known to the Insurer, and the producer of record at least sixty (60) days but not more than one hundred twenty (120) days before the expiration date.
- 2. The notice of nonrenewal will state the actual reason for nonrenewal.
- 3. If notice is mailed, proof of mailing will be sufficient proof of notice.
- 4. A notice of nonrenewal will not be required in any of the following situations:
 - a. The transfer of, or renewal of, a policy without change in its terms or conditions or the rate on which the premium is based between insurers that are members of the same insurance group.
 - **b.** The policy has been extended for ninety (90) days or less, if the notice required has been given prior to the extension.
 - c. The Named Insured has obtained replacement coverage or has agreed, in writing, within sixty (60) days of the termination of the policy, to obtain that coverage.
 - **d.** The policy is for a period of no more than sixty (60) days and the Named Insured is notified at the time of issuance that it may not be renewed.
 - e. The Named Insured requests a change in the terms or conditions or risks covered by the policy within sixty (60) days prior to the end of the policy period.
 - f. The Insurer has made a written offer to the Named Insured, within the prescribed time period, to renew the policy under changed terms or conditions or at a changed premium rate, where the increase is more than 25%. As used herein, "terms or conditions" includes, but is not limited to, a reduction in limits, elimination of coverages, or an increase in deductibles.
- 5. In the case of conditional renewal, failure of the Named Insured to satisfy conditions provided by the Insurer for renewal, by the expiration date of the policy or sixty (60) days after mailing or delivery of such notice, whichever is later, the conditional renewal shall be treated as an effective nonrenewal.

D. CONDITIONAL RENEWAL

- 1. If the policy has been in effect for more than sixty (60) days or if the policy is a renewal, effective immediately no increase in premium, reduction in limits, or change in the conditions of coverage shall be effective during the policy period unless based upon one of the following reasons:
 - a. Discovery of willful or grossly negligent acts or omissions, or of any violations of state laws or regulations establishing safety standards by the Named Insured or Insured(s) which materially increase any of the risks or hazards insured against.
 - **b.** Failure by the Named Insured or Insured(s) to implement reasonable loss control requirements which were agreed to by the Insured as a condition of policy issuance or which were conditions precedent to the use by the Insurer of a particular rate or rating plan, if the failure materially increases any of the risks insured against.
 - **c.** A determination by the commissioner that loss of or changes in an insurer's reinsurance covering all or part of the risk covered by the policy would threaten the financial integrity or solvency of the Insurer unless the change in the terms or conditions or rate upon which the premium is based is permitted.

CNA62814CA (7-20)
Page 2 of 4
CONTINENTAL CASUALTY COMPANY
Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 Endorsement No: Effective Date: 11/22/2022

Cancellation / Nonrenewal - California

- d. A change by the Named Insured or Insured(s) in the activities or property of the commercial or industrial enterprise which results in a materially added risk, a materially increased risk, or a materially changed risk, unless the added, increased, or changed risk is included in the policy.
- 2. A written notice will be mailed or delivered to the Named Insured, at the last mailing address known to the Insurer, and the producer of record at least sixty (60) days prior to the effective date of any increase, reduction or change.
- The notice will state the effective date of, and the reasons for, the increase, reduction or change
- 4. If notice is mailed, proof of mailing will be sufficient proof of notice.

E. ADDITIONAL PROVISIONS

- 1. Solely with respect to coverage for real property used predominantly for residential purposes and consisting of not more than four dwelling units, and to coverage on tenants' household property contained in a residential unit:
 - The Insurer shall not cancel or refuse to renew such coverage existing on the date the Insurer elected to become an associate participating insurer after an offer of earthquake coverage is accepted solely because the insured has accepted that offer of earthquake coverage; and
 - b. The Insurer shall not cancel such coverage unless the policy is properly canceled pursuant to Paragraph A above: and
 - c. The Insurer may not cancel or non-renew this policy solely because the first Named Insured has:
 - (1) Accepted an offer of earthquake coverage; or
 - (2) Cancelled or did not renew a policy issued by the California Earthquake Authority (CEA) that included an earthquake policy premium surcharge.

However, the Insurer shall cancel this policy if the first Named Insured has accepted a new or renewal policy issued by the CEA that includes an earthquake policy premium surcharge but fails to pay the earthquake policy premium surcharge authorized by the CEA.

- d. If the Insurer elects not to renew this policy, the Insurer will mail or deliver written notice, stating the reason for nonrenewal, to the first Named Insured, and to the producer of record, at the mailing address shown in the policy, at least 75 days, but not more than 120 days, before the expiration or anniversary date. The Insurer may elect not to renew such coverage for any reason, except as provided in e., f. and g. below. If the Insurer fails to give the first Named Insured notice of nonrenewal at least 75 days prior to the policy expiration, as required in the paragraph above, this policy, with no change in its terms and conditions, shall remain in effect for 75 days from the date that the notice of nonrenewal is delivered or mailed to the Named Insured. A notice to this effect shall be provided by us to the first Named Insured with the notice of nonrenewal.
- e. The following applies only to insurers who are associate participating insurers as established by Cal. Ins. Code Section 10089.16. The Insurer may elect not to renew such coverage after the first Named Insured has accepted an offer of earthquake coverage, if one or more of the following reasons apply:
 - The policy is terminated by the Named Insured;
 - The policy is refused renewal on the basis of sound underwriting principles that relate to the coverages provided by the policy and that are consistent with the approved rating plan and related documents filed with the Department of Insurance as required by existing law;
 - iii. The Commissioner of Insurance finds that the exposure to potential losses will threaten the solvency of the Insurer or place the Insurer in a hazardous condition. A hazardous condition includes, but is not limited to, a condition in which the Insurer makes claims payments for losses resulting from an earthquake that occurred within the preceding two years and that required a reduction in policyholder surplus of at least twenty-five percent (25%) for payment of those claims; or
 - iv. The Insurer has lost or experienced a substantial reduction in the availability or scope of reinsurance coverage or a substantial increase in the premium charged for reinsurance coverage for its residential property insurance policies, and the Commissioner of Insurance has approved a plan for

Page 3 of 4 CONTINENTAL CASUALTY COMPANY Insured Name: MOFFATT & NICHOL

Policy No: 6056712992 **Endorsement No:**



Cancellation / Nonrenewal – California

the nonrenewals that is fair and equitable, and that is responsive to the changes in the Insurer's reinsurance position.

f. If a state of emergency under California Law is declared and the residential property is located in any ZIP Code within or adjacent to the fire perimeter, as determined by California Law, the Insurer may not cancel or non-renew this policy for one year, beginning from the date the state of emergency is declared, solely because the dwelling or other structure is located in an area in which a wildfire has occurred.

However, the Insurer may cancel or non-renew:

- (1) When the Named Insured has not paid the premium at any time and the Insurer lets the Named Insured know at least 10 days before the date cancellation takes effect;
- (2) If willful or grossly negligent acts or omissions by the Named Insured, or his or her representatives, are discovered that materially increase any of the risks insured against;
- (3) If losses unrelated to the post-disaster loss condition of the property have occurred that would collectively render the risk ineligible for renewal; or
- (4) If there are physical changes in the property insured against, beyond the catastrophe-damaged condition of the structures and surface landscape, which result in the property becoming uninsurable
- g. If this policy contains an exclusion barring coverage for the peril of corrosive soil conditions, the Insurer shall not cancel or refuse to renew the policy solely because corrosive soil conditions exist on the location.

All other terms and conditions of the Policy remain unchanged.

This endorsement, which forms a part of and is for attachment to the Policy issued by the designated Insurers, takes effect on the effective date of said Policy at the hour stated in said Policy, unless another effective date is shown below, and expires concurrently with said Policy.

Policy No: 6056712992



THIS ENDORSEMENT CHANGES THE POLICY. PLEASE READ IT CAREFULLY. EXTENDED COVERAGE ENDORSEMENT – BA PLUS

This endorsement modifies insurance provided under the following:

BUSINESS AUTO COVERAGE FORM

I. LIABILTY COVERAGE

A. Who Is An Insured

The following is added to Section II, Paragraph A.1., Who Is An Insured:

- a. Any incorporated entity of which the Named Insured owns a majority of the voting stock on the date of inception of this Coverage Form; provided that,
 - b. The insurance afforded by this provision A.1. does not apply to any such entity that is an "insured" under any other liability "policy" providing "auto" coverage.
- Any organization you newly acquire or form, other than a limited liability company, partnership or joint venture, and over which you maintain majority ownership interest.

The insurance afforded by this provision A.2.:

- a. Is effective on the acquisition or formation date, and is afforded only until the end of the policy period of this Coverage Form, or the next anniversary of its inception date, whichever is earlier.
- b. Does not apply to:
 - (1) "Bodily injury" or "property damage" caused by an "accident" that occurred before you acquired or formed the organization; or
 - (2) Any such organization that is an "insured" under any other liability "policy" providing "auto" coverage.
- Any person or organization that you are obligated to provide Insurance where required by a written contract or agreement is an insured, but only with respect to legal responsibility for acts or omissions of a person for whom Liability Coverage is afforded under this policy.
- 4. An "employee" of yours is an "insured" while operating an "auto" hired or rented under a contract or agreement in that "employee's" name, with your permission, while performing duties related to the conduct of your business.

"Policy," as used in this provision A. Who Is An Insured, includes those policies that were in force on the inception date of this Coverage Form but:

- 1. Which are no longer in force; or
- 2. Whose limits have been exhausted.

B. Bail Bonds and Loss of Earnings

Section II, Paragraphs A.2.a.(2) and A.2.a.(4) are revised as follows:

- 1. In a.(2), the limit for the cost of bail bonds is increased from \$2,000 to \$5,000, and
- In a.(4), the limit for the loss of earnings is increased from \$250 to \$500 a day.

C. Fellow Employee

Section II, Paragraph B.5 does not apply.

Such coverage as is afforded by this provision C. is excess over any other collectible insurance.

II. PHYSICAL DAMAGE COVERAGE

A. Towing

Section III. Paragraph A.2., is revised to include Light Trucks up to 10,000 pounds G.V.W.

B. Glass Breakage – Hitting A Bird Or Animal – Falling Objects Or Missiles

The following is added to Section III, Paragraph A.3.:

With respect to any covered "auto," any deductible shown in the Declarations will not apply to glass breakage if such glass is repaired, in a manner acceptable to us, rather than replaced.

C. Transportation Expenses

Section III, Paragraph A.4.a. is revised, with respect to transportation expense incurred by you, to provide:

- a. \$60 per day, in lieu of \$20; subject to
- b. \$1,800 maximum, in lieu of \$600.

D. Loss of Use Expenses

Section III, Paragraph A.4.b. is revised, with respect to loss of use expenses incurred by you, to provide:

a. \$1,000 maximum, in lieu of \$600.

E. Personal Property

The following is added to Section III, Paragraph A.4.





ADDITIONAL INSURED - PRIMARY AND NON-CONTRIBUTORY

It is understood and agreed that this endorsement amends the BUSINESS AUTO COVERAGE FORM as follows:

SCHEDULE

Name of Additional Insured Person Or Organization

ANY PERSON OR ORGANIZATION THAT YOU ARE REQUIRED BY WRITTEN CONTRACT OR WRITTEN AGREEMENT TO NAME AS AN ADDITIONAL INSURED.

- 1. In conformance with paragraph A.1.c. of Who Is An Insured of Section II LIABILITY COVERAGE, the person or organization scheduled above is an insured under this policy.
- 2. The insurance afforded to the additional insured under this policy will apply on a primary and non-contributory basis if you have committed it to be so in a written contract or written agreement executed prior to the date of the "accident" for which the additional insured seeks coverage under this policy.

All other terms and conditions of the policy remain unchanged

This endorsement, which forms a part of and is for attachment to the policy issued by the designated Insurers, takes effect on the Policy Effective date of said policy at the hour stated in said policy, unless another effective date (the Endorsement Effective Date) is shown below, and expires concurrently with said policy.

Form No: CNA71527XX (10-2012) Endorsement Effective Date: Endorsement No: Page: 1 of 1

Endorsement Expiration Date:

Policy No: BUA 6056712989 Policy Effective Date: 11/22/2022

Underwriting Company: Transportation Insurance Company, 151 N Franklin St, Chicago, IL 60606





WAIVER OF TRANSFER OF RIGHTS OF RECOVERY AGAINST OTHERS TO US (WAIVER OF SUBROGATION)

THIS ENDORSEMENT CHANGES THE POLICY. PLEASE READ IT CAREFULLY.

This endorsement modifies insurance provided under the following:

AUTO DEALERS COVERAGE FORM

BUSINESS AUTO COVERAGE FORM

MOTOR CARRIER COVERAGE FORM

With respect to coverage provided by this endorsement, the provisions of the Coverage Form apply unless modified by the endorsement.

This endorsement changes the policy effective on the inception date of the policy unless another date is indicated below.

Named Insured: MOFFATT & NICHOL

Endorsement Effective Date: 11/22/2022

SCHEDULE

Name(s) Of Person(s) Or Organization(s):

ANY PERSON OR ORGANIZATION FOR WHOM OR WHICH YOU ARE REQUIRED BY WRITTEN CONTRACT OR AGREEMENT TO OBTAIN THIS WAIVER FROM US. YOU MUST AGREE TO THAT REQUIREMENT PRIOR TO LOSS.

Information required to complete this Schedule, if not shown above, will be shown in the Declarations.

The Transfer Of Rights Of Recovery Against Others To Us condition does not apply to the person(s) or organization(s) shown in the Schedule, but only to the extent that subrogation is waived prior to the "accident" or the "loss" under a contract with that person or organization.

Form No: CA 04 44 10 13 Endorsement Effective Date:

Endorsement Expiration Date:

Policy No: BUA 6056712989 Policy Effective Date: 11/22/2022

Endorsement No: Page: 1 of 1

Underwriting Company: Transportation Insurance Company, 151 N Franklin St, Chicago, IL 60606

WAIVER OF OUR RIGHT TO RECOVER FROM OTHERS ENDORSEMENT

This endorsement changes the policy to which it is attached effective on inception date of the policy unless a different date is indicated below.

This endorsement, effective 12:01 AM 11/22/2022 forms a part of Policy No. WC

19-39-7906

Issued to MOFFATT & NICHOL

By NATIONAL UNION FIRE INSURANCE COMPANY OF PITTSBURGH, PA.

We have the right to recover our payments from anyone liable for an injury covered by this policy. We will not enforce our right against the person or organization named in the Schedule. This agreement applies only to the extent that you perform work under a written contract that requires you to obtain this agreement from us.

This agreement shall not operate directly or indirectly to benefit any one not named in the Schedule.

Schedule

ANY PERSON OR ORGANIZATION WITH WHOM YOU HAVE ENTERED INTO A CONTRACT, A CONDITION OF WHICH REQUIRES YOU TO OBTAIN THIS WAIVER FROM US. THIS ENDORSEMENT DOES NOT APPLY TO BENEFITS OR DAMAGES PAID OR CLAIMED: 1. PURSUANT TO THE WORKERS' COMPENSATION OR EMPLOYERS' LIABILITY LAWS OF KENTUCKY, NEW HAMPSHIRE, OR NEW JERSEY; OR, 2. BECAUSE OF INJURY OCCURRING BEFORE YOU ENTERED INTO SUCH A CONTRACT.

This form is not applicable in Kansas for private construction contracts as defined in K.S.A. 16-1801 through K.S.A. 16-1807 or public construction contracts as defined in K.S.A. 16-1901 through 16-1908, except where permitted by statute or other applicable law, such as for use in wrap-up insurance programs.

Any person or organization for which the employer has agreed by written contract, executed prior to loss, may execute a waiver of subrogation. However, for purposes of work performed by the employer in Missouri, this waiver of subrogation does not apply to any construction group of classifications as designated by the waiver of right to recover from others (subrogation) rule in our manual.

This form is not applicable in California, Kentucky, New Hampshire, New Jersey, Texas, or Utah.

		DEPH. M'SON
WC 00 03 13 (Ed. 04/84)	Countersigned by	

Authorized Representative